

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Blonski, Matthias; von Lilienfeld-Toal, Ulf

Working Paper Excess returns and the distinguished player paradox

cege Discussion Papers, No. 78

Provided in Cooperation with: Georg August University of Göttingen, Department of Economics

Suggested Citation: Blonski, Matthias; von Lilienfeld-Toal, Ulf (2008) : Excess returns and the distinguished player paradox, cege Discussion Papers, No. 78, University of Göttingen, Center for European, Governance and Economic Development Research (cege), Göttingen

This Version is available at: https://hdl.handle.net/10419/32031

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Center for European, Governance and Economic Development Research **Discussion Papers**

Number 78 – October 2008

Excess Returns and the Distinguished Player Paradox

Matthias Blonski Ulf von Lilienfeld-Toal

Georg-August-Universität Göttingen

ISSN: 1439-2305

The article was presented in a lecture within the research colloquium of the Center for European, Governance and Economic Development Research (*cege*) on October 23rd 2008 at the University of Göttingen.

For more information about the Center for European, Governance and Economic Development Research (*cege*) please refer to: <u>http://www.uni-goettingen.de/de/60864.html</u>.

Excess Returns and the Distinguished Player Paradox

Matthias Blonski^{*} Ulf von Lilienfeld-Toal[†]

this draft: 7.8.2008

Abstract

Suppose the value of a firm is endogenously determined by a manager's costly effort. We call this manager a *distinguished player* if he also can trade shares of the firm on a market. Arbitrage-free asset pricing theory suggests that the equilibrium market price reflects the value increasing contribution of a distinguished player. Trade at this price, however, cannot be an equilibrium of a market game since due to private effort costs, shares have a lower value to the distinguished player as compared to other investors. Why? The distinguished player himself can gain by selling at this price and in turn reduce effort. By merging asset pricing and corporate finance concepts we solve this distinguished player paradox and show how this asymmetry in valuations can systematically bring about a trade price strictly below the equilibrium value of the company. This implies that buyers enjoy excess returns on their investment and is thereby at odds with the efficient markets hypothesis. It further involves a substantial reinterpretation of traditional *no-arbitrage* towards a game-theoretic understanding. The empirical prediction that companies with a distinguished player yield excess-returns was confirmed for the sample of S&P500 firms and S&P1500 firms in a companion paper by von Lilienfeld-Toal and Rünzi (2007). Our results are shown to be robust with respect to trading rules, discrete versus continuous effort, trading costs, noise traders, and price taking behavior.

JEL Classification: G12, G32, C72, D43, D46

Keywords: excess returns, underpricing, no-arbitrage, asset pricing, corporate finance

^{*}Department of Economics, Goethe University, Frankfurt am Main, 60054 Frankfurt a.M., Germany; telephone: +49-(0)69-798-23858; e-mail: *blonski@wiwi.uni-frankfurt.de*

[†]Department of Economics, Goethe University, Frankfurt am Main, 60054 Frankfurt a.M., Germany; e-mail: *lilienfeld@wiwi.uni-frankfurt.de*

1 Introduction

Consider a manager of a firm who can raise the value of a firm by exerting costly effort *and* trade shares of the firm on the stock market. Applying the standard no-arbitrage equilibrium concept of asset pricing to determine the value of this firm leads to a paradox. All relevant information should be priced in, including the ownership and thereby the effort of the manager. If, however, the privately costly effort is already priced, the manager would be better off to sell his shares, not exerting effort, and saving the private effort costs instead.¹

This paradox is mirrored in the empirical evidence provided by von Lilienfeld-Toal and Rünzi (2007) who show that standard arbitrage free asset pricing cannot explain the cross section of stock returns for firms with an owner-manager. For example, they show that a value-weighted portfolio consisting of all S&P 500 firms (1994-2005) in which the CEO holds more than 10% of the company's stocks significantly outperforms the total market portfolio by 13% p.a.

In this article we propose a solution to the theoretical paradox that is also consistent with the empirical evidence. We analyze incentives and interactions among completely and symmetrically informed players who can be rational or irrational traders of shares of a company. We introduce a *distinguished player* who can trade *and* increase the firm's value by exerting costly effort. This distinguished player reflects a basic concept of corporate finance – i.e. operates in a standard moral hazard context. For example, the distinguished player could be interpreted as the *agent* in the Grossman and Hart (1983) model.² While the information structure and production technology is similar, we let the distinguished player trade shares – and hence incentives – before exerting effort. The main task is to analyze the outcome of the ex-ante market game before the distinguished player's effort decision.

Two classes of trade equilibria are of interest: *true value* and *excess returns equilibria*. In a true value equilibrium, shares of the firm are traded at the price that equals the equilibrium value of the firm. In excess returns equilibria, shares of the firm are traded at a price strictly below the equilibrium value of the firm.

Our main results are (1) true value equilibria do not exist in rational call auction markets. This formalizes the aforementioned paradox of endogenous firm value and arbitrage free asset pricing. (2) Excess returns equilibria may exist – even if true value equilibria do not exist – which provides an explanation for the abnormal returns of owner-CEO firms as reported in von Lilienfeld-Toal and Rünzi (2007). (3) Excess returns equilibria are robust with respect to (i) trading costs, (ii) noise traders and price taking behavior, (iii) discrete vs. continuous effort and (iv) the specification of the market microstructure. (4) Conversely, excess returns equilibria do not exist without a distinguished player.

¹A related paradox is discussed in Grossman and Stiglitz (1980) where information is costly.

²In this respect we follow the literature on asset pricing with large shareholders (see e.g. Bolton and von Thadden (1998), DeMarzo and Urosevic (2006), or Admati, Pfleiderer and Zechner (1994)).

The main ideas behind these results are now motivated in more detail.

Non-existence of true value equilibria. The result that true value equilibria may not exist³ constitutes the formalization of one part of the paradox and provides a first insight as to why analyzing firms with a distinguished player is an important task that leads to surprising results. The intuition behind non-existence is as follows. Due to private effort costs, the valuation of the distinguished player and outside investors differ. In particular, the distinguished player values shares strictly lower than outside investors. As a result, if shares of the firm are traded at the true value the distinguished player wishes to sell his shares whereas outside investors are indifferent between trading and not trading. Two ingredients are required to show that this cannot be an equilibrium: Continuous effort and anonymous trading. Under continuous effort, the distinguished player always adjusts his effort and hence saves some effort costs if he manages to sell some shares, irrespective of how many shares he sells. This implies that the distinguished player always benefits if he manages to sell without having a significant impact on the share price. Further, anonymous trading makes sure that the distinguished player can indeed sell shares without affecting the price. Since trade is anonymous on real world stock exchanges, this implies that true value equilibria systematically fail to exist if effort is continuous.

Since the non-existence result is a negative result our analysis does not stop here. Rather, we show that stock prices of firms with a distinguished player can still be analyzed within a standard market game since excess returns equilibria exist even in set-ups where true value equilibria do not exist.

Excess returns equilibria. In contrast to standard asset pricing theory "no-arbitrage" in our theory does not imply that the market price and the true equilibrium value coincide. In excess returns equilibria those traders who manage to buy below the equilibrium value do indeed realize *strictly positive gains* even without any informational advantage whereas rational sellers suffer a strict loss. In this sense, the excess returns phenomenon appears to contradict the traditional interpretation of *efficient markets* and *no-arbitrage* in equilibrium (see for example Fama, 1970 or Ross, 1976). The obvious question is apparently, why does not everybody buy maximally at a price for which buying yields strictly positive gains? A game theoretic inspection reveals that in an excess returns *equilibrium* "no-arbitrage" is still valid in the sense that no investor can gain by buying or selling more or less. Rational traders are aware of the fact that shares are traded below their *true value*. But at the same time they acknowledge that the distinguished player has an incentive to sell shares – or buy less – whenever the share price exceeds

³In the market games we consider there always exist equilibria, for example the no-trade equilibrium. Non-existence of true value equilibria only refers to non-existence of this specific type of equilibria, which plays a paramount role in the asset pricing literature.

a certain threshold and this threshold is below the equilibrium value. Hence, trade at the equilibrium value would encourage the distinguished player to sell his shares in an anonymous market and save on effort costs instead. Therefore, in a fully rational environment excess returns equilibria are characterized by the property that any deviation that drives up the market price towards the true value triggers the distinguished player to withdraw instead of raising the company value to the anticipated level which in turn causes even bigger losses to everybody. This latter property of excess returns equilibria is called *pivotalness*. It can serve as a disciplining off-equilibrium coordination device. Put differently, a failure to coordinate on a sufficiently low market price below the true equilibrium value may destroy wealth for all shareholders by removing incentives for the distinguished player to work hard and generate positive externalities.

Since excess returns equilibria can exist and may even be the only equilibria in a standard setting that is extended only by the presence of a distinguished player they attract theoretical attention. Their properties systematically differ from standard no-arbitrage asset pricing models. This allows us to derive new predictions that can be tested in empirical work. For example, this theory correctly predicted the hypothesis tested in von Lilienfeld-Toal and Rünzi (2007). Testing other properties of excess returns equilibria promises to yield further new empirical results that so far have not been investigated, simply for the fact that they are inconsistent with traditional no-arbitrage asset pricing theory. Indeed, excess returns equilibria are consistent with other phenomena that have traditionally been considered as "asset pricing anomalies" (e.g. equity premium *puzzle*, excess volatility, no trade theorems, lockup agreements, limited stock market participation, ...) as is discussed in more detail in von Lilienfeld-Toal (2005).

Robustness: Many small traders and irrationality. Our benchmark analysis is carried out in a fully rational world. Since irrationality and noise traders are considered to be important ingredients of real world stock markets it is unclear so far how robust our results are with respect to the introduction of noise. Moreover, it is unclear whether or not excess returns equilibria survive if the number of investors gets arbitrarily large while the market is noisy. We show that within our *distinguished player environment* neither full rationality and with it pivotalness nor strategic behavior of small traders are necessary ingredients for the existence of a excess returns equilibria. We demonstrate this by analyzing a continuum-trader-version of the model with noise traders. Within the so defined stochastic environment we show existence of excess returns equilibria for several reasonable market mechanisms. The basic idea is the following. With noise final allocations and prices are random variables. As before, the distinguished player plans to sell shares whenever the share price exceeds a certain threshold which now occurs with positive probability. This implies that rational outside investors do not want to buy shares above this threshold because the distinguished player will not exert costly effort in that case. Outside investors only want to buy shares if the share price is below the

threshold price. To make sure not to buy from the distinguished player, outside investors do not submit buy orders with a limit price above the threshold price of the distinguished player. As a result, outside investors do not bid up the share price even though shares are undervalued on average and even though bidding up the share price would be feasible. The existence of these equilibria shows that a distinguished player in combination with irrational noise traders provides an alternative explanation for excess returns equilibria besides pivotalness in fully rational frameworks.

Market microstructure. Our formulation and results establish progress in another direction. It is well known that results in the market microstructure literature hinge critically on the exact specification of the market mechanism. For example, O'Hara (1995) discusses different market microstructure models and states (p. ix) that the "generality of their results, and hence their applicability, is not well understood". This observation is bothering since any specific market mechanism is only an approximation of real world trading systems and it is not known what constitutes a good approximation. The crucial questions are which observations depend on the market mechanism and which don't. Moreover, if they depend on it, how? While our existence results, as the previous literature, depend on the specific market mechanism we also provide characterization results in section 5. They show that some of the unusual predictions on asset pricing with a distinguished player indeed do not depend on the market mechanism. Moreover, our various existence and non-existence results in section 4 are formulated for pricing and allocation rules that are used in the real world. They demonstrate to which extent other details of the relevant equilibria beyond our characterization results in fact do depend on the market mechanism. In particular, we show by comparing different real world exchange rules how the variation of these rules can influence possible equilibrium behavior.

Institutional or contractual clauses. It is difficult to argue that in anonymous markets distinguished players differ from other traders in their ability to commit not to trade. On the one hand, neither SEC regulations nor privately stipulated contracts can fully rule out anonymous trading by distinguished players. The majority of shares held by executives are common shares that are not subject to a non-selling clause. Furthermore, SEC regulations force insiders to report trade ex post and sometimes forces insiders to reveal plans to trade ex ante. Nevertheless, these shares are then traded anonymously on the market and outside investors do not know whether they buy from insiders or from other outsiders. On the other hand, in anonymous markets already the perception of some strategic outside investors about the role of a distinguished player is sufficient to support excess returns equilibria due to this theory.

Large semi-anonymous games. Finally, the multiplicity and rich equilibrium structure of this theory is less surprising from the perspective of *large games* since our model in fact establishes a semi-anonymous game. Semi-anonymity of a game means that players' payoffs only depend on aggregated actions of player-types rather than on the individual action profile. In this context we only consider two types of players, regular investors and the distinguished player. Here, *semi-anonymity* is a crucial property since it means that traders do not care about the composition of bids among regular shareholders. However, being of a different type, the distinguished player's actions distinctly enter the preferences of traders. This very circumstance contributes to bring about the pattern of true value versus excess returns equilibria. Equilibria of semi-anonymous games in general have been characterized in Blonski (2005). Strong theoretical support for the so identified equilibrium structure in these games comes from Kalai (2004) who has shown that equilibria of semi-anonymous games are surprisingly robust in various senses when the number of players gets large.

The paper proceeds as follows. Section 2 introduces the main idea within a simple illustrative example. Section 3 establishes the reference model, introduces formally the idea of a distinguished player, and sets up the notation for general market mechanisms. Any particular market mechanism specifies a corresponding market game. In section 4 we show non-existence of true value equilibria and existence of excess returns equilibria for several real world call auctions and the Kyle (1989) market mechanism. In section 5 we characterize excess returns and true value equilibria for general market mechanisms. Section 6 derives robustness results and establishes existence of the excess returns equilibrium with a continuum of traders and noise traders. Section 7 discusses in more detail the relationship of this theory to the most relevant existing literature. Section 8 discusses further extensions like multiple distinguished players or a market maker while section 9 concludes. Appendix A explains more rigorously the rich and general strategy space of the market game, Appendix B introduces the language for stochastic market mechanisms while Appendix C contains all proofs which are not in the main text.

2 Intuitive Example

Consider four players $i \in I = \{0, 1, 2, 3\}$ who jointly own a business project. Suppose for simplicity that initial ownership of the project is subdivided into four indivisible shares of equal size $(\alpha_0, \alpha_1, \alpha_2, \alpha_3) = (1, 1, 1, 1)$. Imagine that players i = 1, 2, 3 are wealthy in contrast to player i = 0. However, player i = 0 – called the *distinguished player* – is a genius who once upon a time has a brilliant idea how to raise the value of the project from $\underline{v} = 0$ to $\overline{v} = 40$. However, to implement and materialize this idea the distinguished player has to work hard and exert effort $e \in \{0, 1\}$ facing private effort cost c(e) = 4e. Finally, the project is sold for its terminal value and each share is worth 10e. Everything is public information. Without trade this world is quite trivial, the distinguished player certainly exerts high effort e = 1 being aware of the fact that the final value of his share exceeds his private effort cost. The final value of each share is 10 and payoffs are $(u_0, u_1, u_2, u_3) = (6, 10, 10, 10).$

However, this was just the background story. The main object of interest is a simple anonymous market for stakes of the project before the distinguished player decides on effort. The role of players i = 1, 2, 3 in this example is to perform a very simple version of this anonymous market with noise. Players i = 1, 2 are rational traders while player i = 3 behaves "irrationally" and trades for some exogenous reason, for example a liquidity shock. Instead of friends or business partners we now imagine nameless anonymous shareholders.

Market game. While the rules of the market game in this example are specific and simple they already display some properties of real world stock markets as anonymity, trade volume maximization and price priority. Every player *i* simultaneously can either do nothing or announce one order. This order can either be a buy order of quantity 1 at price $p_i^b \in \{0, 1, \dots, 10\}$ or a selling order of quantity 1 at price $p_i^s \in \{0, 1, \dots, 10\}$. The distinguished player in this example cannot afford to buy. Therefore, i = 0 can only submit a sell order or do nothing. The market is assumed to clear as follows. There is trade if and only if at least one buying order p_i^b and one selling order p_j^s are submitted such that $p_j^s \leq p_i^b$. If there are more than one competing buying orders with different limit prices and one selling order, such that all buying prices are at least as high as the selling price only the order with the higher price is executed against the selling order. If the buying order prices coincide each of them is executed with equal probability. Correspondingly, if there are one buying order and more than one selling orders with limit prices below the buying price only the lower selling order is executed, or again, if identical all are executed with equal probability. Finally, if there is more than one price maximizing the trade volume the market mechanism picks the lower price.

The noise trader i = 3 is assumed to do nothing with probability $1 - \lambda$, to submit a sell market order $p_3^s = 0$ with probability $\frac{\lambda}{2} \ge 0$ and to submit a buy market order $p_3^b = 10$ with probability $\frac{\lambda}{2}$ (negative and positive liquidity shocks).

Non-existence of the true value equilibrium. We first show that there exists no trade equilibrium at the high equilibrium value $p^* = 10$. If there were such an equilibrium there must be a buy offer of a non-distinguished player i = 1, 2, 3 with $p_i^b = 10$ and any sell offer $p_i^s = 10$ both with strictly positive probability. However, this cannot be an equilibrium since in this case the distinguished player can improve by submitting a sell order at $p_0^s = 9$ and only to exert high effort if his sell order is not executed. This would yield an expected market price p^* strictly below 10 which contradicts $p^* = 10$.

Excess returns equilibrium. Is there any trade equilibrium at an expected lower price? The answer is: Yes, for example at $p^* = 6$. To understand this equilibrium first note that for the distinguished player it can be a decent strategy to submit a sell order

at price $p_0^s = 7$ since any realized sell for a price larger than 6 can only raise his payoff compared to the payoff of not trading which is 6. Now life gets troublesome for the bunch of rational anonymous traders who are unable to commit. If the distinguished player were to succeed in selling his share, he would in turn not work anymore and there would be nothing to gain of his brilliant idea. The rational traders i = 1, 2 realize that the noise trader i = 3 buys at any price with probability $\frac{\lambda}{2}$. A trader who recognizes this, say trader i = 1, may wish to make sure that at least the worst case will be avoided after which all shares would end up to be worth 0. A possibility to avoid this worst case is to submit a sell order at $p_1^s = 6$ below the price limit of the distinguished player. Now, from the perspective of player i = 2 at this low price it is clearly attractive to buy if one can be sure not to buy from the distinguished player, for example by submitting a buying order $p_2^b = 6$. It is not difficult to see that this order profile indeed establishes an equilibrium for appropriate values of noise λ . First note that equilibrium price is $p^* = 6$. There are two potentially critical non-deviation constraints. (i) Player i = 2 should not be able to gain by bidding up the expected price, for example by submitting instead $p_2^b = 10$, and (ii) player i = 1 should not gain by withdrawing the selling order $p_1^s = 6$. In case (i) the distinguished player gets his selling order executed with probability $\frac{\lambda}{2}$, namely, when the noise trader also submits $p_3^b = 10$. While in this state of nature player 2's payoff gets negative and decreases in every other state i = 2 is the only buyer and market price remains $p^* = 6$ and hence 2's payoff unchanged. To see that case (ii) does not establish a beneficial deviation for trader i = 1 for appropriate values of λ consider the three possible states of nature (1) noise trader i = 3 buys and (2) sells each with probability $\frac{\lambda}{2}$ and (3) noise trader does not trade with probability $1 - \lambda$. In state (1) deviation (ii), i.e. withdrawing the selling order triggers the bad event that the distinguished player sells to the noise trader and 1's payoff drops from 6 to 0. In state (2), however, player i = 2buys from the noise trader and the distinguished player keeps on working hard. In this case 1's payoff rises from 6 to 10. In state (3) withdrawing the selling order yields no trade and thereby again 1's payoff rises from 6 to 10. To sum up, by withdrawing the sell order raises 1's payoff 10 - 6 = 4 units with probability $1 - \frac{\lambda}{2}$ in states (2) and (3) but decreases 1's payoff 6 - 0 = 6 units with probability $\frac{\lambda}{2}$ in state (1). Together this deviation is not beneficial for

$$4 \cdot \left(1 - \frac{\lambda}{2}\right) \le 6 \cdot \frac{\lambda}{2}$$

or $\lambda \geq \frac{4}{5}$. In other words, rational player 1 prefers to sell at price p = 6 below the equilibrium value 10 to avoid the case that the distinguished player sells his share and the value drops to 0 if noise trading renders the latter case sufficiently likely without this selling order. The bitter truth for rational player i = 1 is therefore that selling at a price 6 below the anticipated value is the only way to keep the distinguished player from selling considering that the project's value fully hinges on the distinguished player's abilities.

It is easy to check that there are further equilibria. Clearly, the roles of the *equilibrium* winner i = 2 and *equilibrium looser* i = 1, i.e. the players who realize strict gains and

losses relative to the equilibrium value by their trading behavior, may be permuted. However, there are also other equilibrium prices, all of them strictly below equilibrium value. Since pursuing this further within this example does not add much further insight we sum up the main observations and move on to the general framework.

There exists no *true value equilibrium*, i.e. a trade equilibrium where shares of the project are traded at their correctly anticipated equilibrium value 10 while there exist *excess returns equilibria* where buyers enjoy excess returns on their investment. Generally, the presence of a distinguished player with the associated payoff externalities is inconsistent with crucial concepts of capital market theory, the efficient markets hypothesis and traditional no-arbitrage. Available and even public information about a project's value is not reflected by the trade price in an anonymous market. We will learn in more detail, how the market mechanism has to provide some anonymity. If traders could identify their trading partners it is easy to imagine a true value equilibrium where nobody buys from the the distinguished player. Although some details of this example hinge on its specification the remainder of this article shows that all crucial observations are indeed much more general.

3 Market Game with a Distinguished Player

Distinguished player. Denote by i = 0 a distinguished player being interpreted as a "manager-owner", raider, activist shareholder, or founder of a firm. Further, denote by i = 1, ..., N outside investors with (weakly) positive stakes in this firm⁴. Distinguishedness of a player-investor is defined as the ability to enhance the value of the firm. The firm either has value \underline{v} or $\overline{v} \equiv \underline{v} + \Delta v \geq \underline{v}$ and the realization of this value depends on the distinguished player's effort decision which can be discrete $e \in \{0, 1\}$ or continuous $e \in \mathbb{R}_+$. If the distinguished player exerts effort e the firm yields value $v = \underline{v} + e(\overline{v} - \underline{v})$. Exerting effort causes private effort costs $c(e) = c \cdot e^2$. We assume that $\Delta v > c > 0$ so that the efficient or first best effort choice is $e^{FB} > 0$, in both the continuous and the discrete effort case. To compare with known models in the literature, we are also interested in the case $\Delta v = \overline{v} - \underline{v} = 0$ where no player is a distinguished player.

Firm ownership. Our object of investigation is a market game where stakes of the firm can be traded before the distinguished player decides on his effort decision. The *initial ownership structure* of the firm before the market game takes place is exogenously given. It is defined by an element α of the simplex

$$\Delta = \left\{ \xi = (\xi_0, ..., \xi_N) \, \middle| \, \xi_i \in Q \text{ and } \sum_{i=0}^N \xi_i = 1 \right\}$$

 $^{^{4}}$ To study the role of small investors and price taking behavior we study a continuum of investors in section 6.

where quantities Q can be either continuous Q = [0, 1] or discrete $Q = \left\{0, \frac{1}{M}, \frac{2}{M}, ..., \frac{M}{M}\right\} \subset [0, 1]$ with M indivisible shares. Within the latter interpretation, initially player i owns $\alpha_i M$ shares of the firm. The market game to be described subsequently endogenously results in the final ownership denoted by $\omega = (\omega_0, \omega_1, ..., \omega_N) \in \Delta$ with $\omega_i \in Q$ and $\sum_{i=0}^{N} \omega_i = 1$.

Effort choice. In the market game yet to be defined, stakes of the firm are traded before the distinguished player decides about his effort. Once the market game is over the distinguished player chooses effort to maximize the net value

$$\omega_0 \left(\underline{v} + e\Delta v \right) - c(e)$$

of his final stake ω_0 in the firm. Let

$$e(\omega) \in \operatorname*{argmax}_{e} \omega_0 \left(\underline{v} + e\Delta v \right) - c(e).$$
(1)

For discrete effort the distinguished player may be indifferent between 0 and 1. For this case we suppose as the breaking rule that effort is 1. Similarly, the payoff of any outside investor i = 1, ..., N after the market game is given as $\omega_i (\underline{v} + e\Delta v)$, i.e. the final value of his stake after the distinguished player's effort choice.

We assume effort to be non-contractible and that no additional contractual provisions are used. This approach is also used in the related literature, see e.g. Admati, Pfleiderer and Zechner (1994) or DeMarzo and Urosevic (2006) and is supported by empirical evidence.⁵ We believe that as long as the market price enters the distinguished player's incentives, the problems we discuss here cannot be perfectly alleviated by appropriately chosen contractual clauses. We therefore simply assume that no contractual provisions are used at all.

Prices and strategies. To allow for models with discrete or continuous prices we introduce $P \subset \mathbb{R} \cup \{-\infty, \infty\}$ as the set of feasible prices. Similarly as for quantities the relevant examples are a continuous $P = \mathbb{R} \cup \{-\infty, \infty\}$ or a discrete price range $P = P_{\delta} := \{-\infty, ..., \underline{v}, \underline{v} + \delta, \underline{v} + 2\delta, ..., \overline{v}, ..., \infty\}$ with some exogenous tick size δ . Real world market mechanisms distinguish between buy and sell prices p_b, p_s . The difference $\gamma := p_b - p_s \ge 0$ is called *bid ask spread*⁶. If prices $p \in P_{\delta}$ are discrete clearly the bid ask spread γ is supposed to be a non negative integer multiple of the tick size, i.e.

⁵According to the execucomp database, there are 5106 officer year observations, where an officer owns more than 5% of unrestricted shares within all S&P 500 or S&P 1500 firms between 1992-2004. In contrast, only in 26 officer years an officer holds more than 5% of restricted shares. For a description of the data, see von Lilienfeld-Toal and Rünzi (2007).

⁶We explicitly want to include the case of bid ask spread $\gamma = 0$ to be able to compare this model with models without trading costs.

 $\frac{\gamma}{\delta} \in \mathbb{N}$. From here $p \equiv p_s$ always denotes sell prices whereas the corresponding buy price is $p_b = p + \gamma$. Clearly, all buy orders depend on p_b and all sell orders depend on p_s .

Strategies or market actions $a_i \in A_i$ of an investor i are collections of buy and sell orders. Mathematically, a strategy a_i can be described by a pair $\{D_i(p), S_i(p)\}$ of set-valued demand and supply functions or correspondences. For example, $q \in S_i(p)$ represents a quantity of shares trader i is willing to sell at price p. Together, $Z_i(p) = D_i(p) - S_i(p) \subset Q$ is a set of positive or negative net quantities composed by demand and supply quantities that would be acceptable for investor i at price p.⁷ In appendix A we describe in more detail how the rich mathematical object of an excess demand correspondence is the result of a general and realistic set of possible orders such as buy and sell limit orders, market orders, stop orders and all or nothing orders (fill or kill orders). Further, appendix C contains existence proofs for specific market mechanisms and displays pictures with aggregated excess demand correspondences that illustrate the structure of relevant strategy spaces.

Market mechanism. By adding up individual behavior an action profile $a \in A$ induces the market excess demand correspondence

$$Z(p) = \sum_{i=0,\dots,N} Z_i(p)$$

which decomposes into aggregated buy and sell offers

$$D(p) = \sum_{i=0,...,N} D_i(p) \text{ and } S(p) = \sum_{i=0,...,N} S_i(p)$$

called the market demand and market supply correspondences. They define sets of quantities the market as a whole is willing to buy or to sell at a given sell price p. Relevant for many real and theoretical market mechanisms is the *limit order trade volume* $\tau(p)$ for $p \in P$ defined by the maximum tradable quantity

$$\tau(p) = \min\left\{\sup\sum_{i=0,\dots,N} d_i(p), \sup\sum_{i=0,\dots,N} s_i(p)\right\}$$

of the short side of the market restricted to limit orders and stop orders⁸ but excluding all-or-nothing orders.

Denote by $\phi = (x, y) = ((x_0, ..., x_N), (y_0, ..., y_N))$ a buy-sell-transaction vector and $\Phi = \left\{ \phi = (x, y) \mid x_i \in Q, y_i \in Q, \sum x_i - y_i = 0, \alpha + x - y \in \Delta \right\}$

⁷To allow traders as in reality to choose demand and supply rather than just the sum of both - i.e. excess demand - opens the possibility for "beller strategies" in which a trader might, for example, try to bid up the stock price by submitting buy orders and simultaneously selling stocks. It turns out that these strategies complicate existence proofs but we want to consider them since they are not ruled out in many real world trading systems.

⁸The small letters d_i and s_i indicate convex valued demand and supply correspondences composed only by limit orders and stop orders. For details see appendix A on page 31.

the set of feasible buy-sell transaction vectors and x - y a corresponding net trade vector. For initial allocation α and net trade vector x - y the final allocation is $\omega = \alpha + x - y \in \Delta$.

We call player *i* strictly wealth constrained iff *i* can only submit sell orders⁹. For a wealth constrained player *i* a bid strategy consists only of selling bids and therefore takes the form $a_i = \left\{ (\sigma_i^1, \sigma_i^2, ...), (\mathring{\sigma}_i^1, \mathring{\sigma}_i^2, ...), (\psi_i^1, \psi_i^2, ...), (\mathring{\psi}_i^1, \mathring{\psi}_i^2, ...) \right\}$ or $z_i(p) = -S_i(p)$.

Definition 1 For any initial ownership $\alpha \in \Delta$ and any strategy profile $a \in A$ a deterministic market mechanism μ with bid ask spread γ is a mapping

$$\mu : \Delta \times A \to P \times \Phi$$
 with $\mu(\alpha, a) = (p^{\mu}(a), x^{\mu}(a), y^{\mu}(a))$

where for initial ownership α and strategy profile a the market mechanism μ picks a sell price $p^{\mu}(a) \in P$, a buy price $p^{\mu}(a) + \gamma$ and for any player a subset of submitted orders being executed, i.e.

$$x_i^{\mu}(a) \in D_i(p) \cup \{0\}$$
 and
 $y_i^{\mu}(a) \in S_i(p) \cup \{0\}.$

We call this latter property "voluntary trade". Hence, net trades $x_i^{\mu}(a) - y_i^{\mu}(a) \in Z_i(p) \cup \{0\}$ are composed by submitted orders. By specifying the trade vector the market mechanism μ thereby determines the expost ownership structure given as

$$\omega^{\mu}(a) := \alpha + x^{\mu}(a) - y^{\mu}(a).$$

Voluntary trade $x_i \in D_i(p) \cup \{0\}$, $y_i \in S_i(p) \cup \{0\}$ means that only submitted orders or nothing are executed, i.e. nobody can be forced to trade and conversely nobody can enforce trade. Market mechanism μ is said to maximize the trade volume if μ picks a price that maximizes the limit order and stop order trade volume $\tau(p)$. Of course, market mechanisms can be specified that do not maximize the trade volume. For example, price could be determined by maximizing the total trade volume including fill or kill orders. Alternatively the distinguished player or other players could be treated with priority if they submit orders. However, we are not aware of a real world market mechanism not using maximal trade volume with top priority. Nevertheless, our characterization results in section 5 do not rely on this property of maximal trade volume. Conversely, we will see in Proposition 2 that a property like maximal trade volume may be sufficient to guarantee existence of trade equilibria. Moreover, a market mechanism is called anonymous if permuting players' names does not affect price and final allocation.

Market game Γ_{μ} . Any market mechanism μ with bid ask spread γ together with an initial ownership α induces a *market game* Γ_{μ} with strategy space A and *payoff functions*

⁹For example, this is likely to be the very reason why the distinguished player needs funding by outsiders. Otherwise he would prefer to run the firm himself.

given by

$$u_i(a) = \omega_i^{\mu}(a) \left(\underline{v} + e(\omega^{\mu}(a))\Delta v\right) - \left(p^{\mu}(a) + \gamma\right) x_i^{\mu}(a) + p^{\mu}(a) \cdot y_i^{\mu}(a)$$

for i = 1, ..., N and

$$u_0(a) = \omega_0^{\mu}(a) \left(\underline{v} + e(\omega^{\mu}(a)) \cdot \Delta v\right) - \left(p^{\mu}(a) + \gamma\right) x_0^{\mu}(a) + p^{\mu}(a) \cdot y_0^{\mu}(a) - c(e(\omega^{\mu}(a)))$$

for the distinguished player i = 0. Any action profile a in a deterministic market mechanism generates a unique final allocation $\omega^{\mu}(a)$ and thereby induces a unique optimal effort decision $e(\omega^{\mu}(a))$ as derived in equation (1). Hence, company value v(a) is given as

$$v^{\mu}(a) = \underline{v} + e(\omega^{\mu}(a))\Delta v$$

No arbitrage. A strategy profile a of the market game Γ_{μ} is said to satisfy the traditional *no-arbitrage condition* iff there exists no trader i who realizes strictly positive gains under the price $p^{\mu}(a)$ and buy-sell transactions $x_i^{\mu}(a)$, $y_i^{\mu}(a)$ determined by market mechanism μ :

$$[p^{\mu}(a) - v(a)] y_{i}^{\mu}(a) + [v(a) - p^{\mu}(a) - \gamma] x_{i}^{\mu}(a) \le 0$$

for all $i = 0, \ldots, N$.

Generalized No-arbitrage and Equilibrium. A strategy profile a^* is said to satisfy generalized no-arbitrage if it is a Nash equilibrium or just equilibrium of market game Γ_{μ} , i.e. no player can strictly improve or in the language of game theory every player plays a best response a_i^* to other players strategy profiles a_{-i}^* . Correspondingly, $(p^*, x^*, y^*) = \mu(\alpha, a^*)$ and $\omega^* = \alpha + x^* - y^*$ are called equilibrium price, equilibrium trades and equilibrium ex post allocation of market game equilibrium a^* under market mechanism μ . Furthermore, we will call $e(\omega(a^*))$ equilibrium effort denoted by e^* . While any equilibrium by definition satisfies generalized no arbitrage hitherto it is not clear if the same holds for traditional no-arbitrage.

No Trade Equilibrium. If no player submits an order no player can gain anything by submitting orders. This simple observation together with the *voluntary trade property* guarantees that for any market mechanism there always exists a no-trade equilibrium where no player submits orders or trades. In the remainder of this article we concentrate on more interesting equilibria where we can observe a price such that trade occurs.

True value and excess returns equilibria. An equilibrium a^* with $\omega^* \neq \alpha$ is called a *trade equilibrium* of Γ_{μ} . *Excess returns* for a firm are defined as

$$R^{\mu}(a^*) := v^{\mu}(a^*) - p^{\mu}(a^*),$$

i.e. the difference between equilibrium firm value and equilibrium price.

Definition 2 A trade equilibrium in which shares are traded at their equilibrium value is called a *true value equilibrium*, i.e.

$$v^{\mu}(a^{*}) = p^{\mu}(a^{*}) \Leftrightarrow R^{\mu}(a^{*}) = 0 \text{ and } \omega^{*} \neq \alpha.$$

A trade equilibrium in which shares are traded below their equilibrium value is called an *excess returns equilibrium*, i.e.

$$v^{\mu}(a^{*}) > p^{\mu}(a^{*}) \Leftrightarrow R^{\mu}(a^{*}) > 0$$
 and $\omega^{*} \neq \alpha$.

A trade equilibrium with high equilibrium effort $e^* > 0$ and equilibrium price $p^* > \underline{v}$ is called a *high* true value equilibrium and a true value equilibrium with low effort $e^* = 0$ and equilibrium price $p^* = \underline{v}$ is called *low* true value equilibrium.

Excess returns equilibria. In an excess returns equilibrium it appears more attractive to be a buyer than a seller. A net equilibrium buyer i who buys enough to strictly overcome transaction cost losses gains

$$(v^* - p^* - \gamma)x_i^* - (v^* - p^*)y_i^* > 0$$

and is called *equilibrium winner*. Although the role of its counterpart – the net equilibrium seller – is less pleasant it can well be rational if the alternative is low effort of the distinguished player triggering a lower value for all.

4 Trading Rules and Existence.

In this section we show existence of excess returns equilibria and non-existence of true value equilibria. To show existence and non-existence, we need to consider any potential deviation for every trader. Since market rules determine how the market price and the allocation changes due to a change of submitted orders, we must become very explicit concerning market rules. In contrast to this, our characterization results hold for general market micro-structures. These characterization results show that the economic intuition behind our results are robust and carry over to a significantly more general class of market microstructures.

To show existence, we concentrate here on electronic call auctions, because 1) we can use the exact and fully specified rules taken from real world trading mechanisms, 2) we do not need to specify the timing as to who trades when and knows what, and 3) these rules are used in the real world and in the literature.¹⁰

¹⁰Trading rules can be ordered by continuous trading vs. call auctions and electronic market places vs. dealer markets. For existence and non-existence results, one cannot use dealer markets since they

While we limit our analysis of existence on real world call auctions, there is still considerable degree of freedom in the details. Price setting rules are less problematic since the same rules are used in almost all call auction mechanisms on stock exchanges around the world. In contrast, rationing rules regulating allocation and the decision which orders are executed must be treated with great care. Rationing rules differ across exchanges and are crucial for our analysis, in particular rules concerning *size priority* and *fill-or-kill orders*.¹¹

Size priority. Size priority is used in different exchanges¹² and in one example we use an opening auction motivated by the Tokyo stock exchange. The trading period in Tokyo starts with an opening auction and orders valid for the opening auction can be submitted prior to the trading period. Furthermore,

"...all limit orders received prior to the start of trading have equal time priority. Price, time, and size priority hold in this order for limit orders placed during the trading sessions." (Lehmann and Modest, 1994, p. 954)

Breaking up orders. Different priority rules are applied at NYSE.

"The NYSE does not follow a strict time priority rule. To minimize the breaking up of large orders, the time priority rule applies only to the first limit order. The remaining limit orders follow a size priority rule; namely limit orders that match the size of the market order at the best price are given priority over other limit orders ..." (Huang and Stoll, 2001, p. 506)

Accordingly, the objective of not breaking up orders is applied in the market mechanism we call NYSE.¹³

Fill-or-kill orders. Similar to size priority, fill-or-kill orders are allowed at some – e.g. Amsterdam or XETRA – but not all real world call auctions. Note, however, that they

¹¹Reny and Perry (2006) put similar emphasis on the exact specification of rationing rules.

¹²There are also several exchanges where size priority is not used, for example at the Paris Bourse (see Biais et al. (1995) or Australia stock exchange (see Aitken et al. (1998)).

¹³It can be argued that the upstairs market used at NYSE and many other exchanges – e.g. Paris Bourse or XETRA – also gives priority to large orders since only large orders can be traded upstairs (and also downstairs).

are not sufficiently explicit. For example, as a rule specialists on the NYSE "... have an exchange mandated obligation to maintain fair and orderly markets." (Lehmann and Modest (1994, p. 952)). To show existence, it is necessary to overcome the lack of preciseness in the regulation of specialists. In the literature, this problem is often solved by assuming that there is perfect competition between market makers and hence equilibrium price equals equilibrium value. Since a major goal of this paper is to derive the equilibrium trade price endogenously and to allow for a deviation from this assumption, this approach is not feasible here. Moreover, continuous trading is more involved as we would have to specify the timing of orders and the information set for every agent.

can be used in most continuous trading settings, e.g. NYSE or Paris¹⁴.

To establish a realistic setting we define four different market micro-structures which are all call auctions. All four call auctions use the same price setting rules and set price priority as the most important allocation rule. Rationing rules apply if price priority does not already lead to a unique allocation. It is at this point, i.e. the second rule concerning allocation, that these market mechanisms differ.

Example 1: Call auctions.

A: Price setting.

- 1. The price is set to maximize the trade volume $\tau(p)$.¹⁵
- 2. Should there be more than one such price, surplus $|s^*(p) d^*(p)|$ is minimized, not counting fill-or-kill orders.
- 3. Should there still be more than one potential price, the minimal price will be taken if there is excess supply. For excess demand, the maximum price is taken.
- 4. Should there still be more than one price, the price closest to a reference price will be chosen and we choose \bar{v} to be the reference price.¹⁶

¹⁶The reference price in real world trading systems is the last traded price (XETRA, p. 27). Since our model only allows for one round of trading, we cannot use the last traded stock price as the reference price.

¹⁴See Venkataraman (2001, p. 1450).

¹⁵Fill or kill orders can be submitted. However, they do not have an impact on price setting. This means that the price and the corresponding executable trading volume or excess demand are calculated as if the fill-or-kill order was not present. A description of the Amsterdam stock exchange (AON are all or nothing orders which is another word for fill or kill orders) as taken from http://www.keytradebank.com/form.html?level=form&option=rul&market=aex is similar: " on the segment of the double auction, ... the fixing price is calculated without the AON orders. Just before the fixing, the AON orders are added to the orderbook."

B: Allocation rules:

Rule	Amsterdam μ_A	Tokyo μ_T	NYSE μ_N	AbsoluteSizePriority μ_S
i.)	Orders are executed according to price priority. This rule does not apply to fill-or-kill orders. Stop orders are not executed.			
ii.)	Fill or kill orders are only matched against each other if they cannot be executed against normal bids. The al- location of fill-or-kill orders maximizes executable trad- ing volume.	Fill-or-killor-dersarenotexecuted.Or-derswithlimitprice p^* areexecutedusingsizepriority.	Fill-or-kill orders are not executed. Fully executable orders using p^* as limit price are executed first. ¹⁷	Orders with limit price p^* and all fill-or-kill orders are exe- cuted according to size priority.
iii.)	Orders with the same priority are executed in a random order.			

The second set of rules, our reinterpretation of the Kyle (1989) market microstructure¹⁸ was chosen because this market microstructure is prominent in the literature. This allows to compare our results with the results of other papers building on the Kyle market microstructure.

Example 2: Kyle market mechanism μ_K . Our reformulation of the Kyle microstructure is as follows:

- 1. The price is set to maximize the trading volume and
- 2. the quantity allocation maximizes trading volume.
- 3. Among those prices and quantity allocations obeying (i) and (ii), Kyle's (1989) market mechanism picks a price p that minimizes absolute value. If both p and -p satisfy this property the market mechanism picks the positive price.
- 4. Market transaction vector ϕ minimizes $\sum_{i} (x_i)^2 + (y_i)^2$.
- 5. Market transaction vector ϕ does not execute fill or kill orders¹⁹.

¹⁹Actually Kyle does not allow fill or kill orders which is equivalent to not being executed. This assumption is necessary to guarantee convex valued excess demand correspondences. Our results, however, do not rely on this assumption.

¹⁷Think of this rule as follows: Every order on the short side of the market and every order on the long side of the market which does not use p^* as limit price are matched first. From the remaining orders on the long side of the market, an order is drawn from the subset of all executable orders. After this draw has been matched, another order is drawn from the (new) subset of fully executable orders. This procedure is continued until no fully executable order exists on the long side of the market. Then, a draw is taken from all remaining orders that use p^* as the limit price and this order is broken up.

¹⁸Actually, Kyle (1989) did not mention trade volume maximization. However, without this latent assumption and market transactions minimization the market mechanism would always implement no trade.

Existence. We are now in a position to formulate our existence and non-existence results.²⁰

Our first result is negative. True value equilibria do not exist in a fully rational market if effort is continuous and trade is organized in anonymous call markets. Our second result is positive. In the same set-up excess returns equilibria exist and we can thereby analyze stock price behavior of firms with a distinguished player using the standard equilibrium concept in a fully rational framework. Excess returns equilibria exist for both, continuous or discrete effort.

Our interpretation of existence and non-existence results at this point is as follows. If we are interested in a rational theory of asset pricing with distinguished players, it is crucial to understand the properties of excess returns equilibria as they may be the only rational equilibria that exist. Furthermore, the pricing predictions from our existence results have been taken to the data by von Lilienfeld-Toal and Rünzi (2007) who find evidence for excess returns equilibria for firms with owner-CEOs. This provides a second motivation for the analysis of excess returns equilibria.

Theorem 1 Consider the market game Γ_{μ} with sufficiently small tick size δ . Suppose effort is continuous $e \in \mathbb{R}_+$ and tradable quantities are discrete. Then, the following is true.

- (I) There exists no true value equilibrium under the Amsterdam, Tokyo, and NYSE market microstructure for zero bid ask spread $\gamma = 0.2^{11}$
- (II) However, there exists an excess returns equilibrium under Amsterdam, Tokyo, NYSE, absolute size priority, and Kyle market microstructure for zero bid ask spread γ = 0.

Theorem 2 Consider the market game Γ_{μ} with sufficiently small tick size δ . Suppose effort is discrete $e \in \{0, 1\}$ and consider a sufficiently small bid ask spread γ and let prices be discrete.

(A) Small initial ownership α_0 . Let i = 0 be a distinguished player with initial stake $\alpha_0 < \frac{c}{\Delta v}$. Then, an excess returns equilibrium exists under any trade volume maximizing market mechanism. Furthermore, an excess returns equilibrium with an equilibrium winner exists under any call auction mechanism (Amsterdam, Tokyo, NYSE, absolute size priority, and Kyle) if $(1 - \frac{1}{M}) \cdot \Delta v \ge c$.

²⁰Note that the call auctions we use here are stochastic market mechanisms. The rigorous formulation of stochastic market mechanisms and the implied stochastic market game is postponed to appendix B and will be used extensively in the proofs in appendix C.

²¹If $\gamma > 0$ true value equilibria do not exist for a more general class of market microstructures unless investors are pivotal which is shown in section 5.

- (B) High initial ownership α_0 . Let i = 0 be a distinguished player with initial stake $\alpha_0 \geq \frac{c}{\Delta v}$. Then, there are initial ownership structures α such that an excess returns equilibrium with an equilibrium winner
 - (i) exists in the Amsterdam Market mechanism μ_A if the distinguished player i = 0is strictly wealth constrained and
 - (ii) exists for the Kyle Market Microstructure μ_K .

All proofs are to be found in appendix C, page 33ff with the exception of the proof for discrete effort and small initial ownership which follows below in the main text.

Theorem 1 shows that trade at the true value is inherently instable. The distinguished player always wants to sell shares. If effort is continuous, selling one single share already yields an improvement for the distinguished player.

Under the same conditions, however, there exist excess returns equilibria. In this respect, trading strictly below equilibrium value is more robust since excess returns equilibria are the only to exist if effort is continuous and all investors are rational. Note that it is straightforward to construct equilibria with substantial excess returns under the Amsterdam, or absolute size priority market microstructure. Under the Amsterdam market microstructures, for example, it suffices if the distinguished player is the only buyer who submits a fill or kill order buying a large number of shares.

The non-existence proof proceeds by characterizing any potential candidate true value equilibrium and finding a contradiction. In every candidate true value equilibrium the following is true. Either the distinguished player has an incentive to change his ex post holding and adjust effort accordingly which is feasible if effort is *continuous*. Or outside investors have an incentive to change their ex post holdings to trade less against the distinguished player. The driving force is that outside investors and the distinguished players can never be indifferent at the same time between buying and selling due to different valuations of shares of the firm. Anonymity of our call auctions makes sure that outside investors or the distinguished player can indeed deviate and change ex post ownership ω .

The intuition behind the existence result of excess returns equilibria is similar for discrete and continuous effort choice. To build up this intuition we proceed with the existence proof for small initial ownership under discrete effort choice.

PROOF (OF THEOREM 2, PART (A)) Suppose discrete quantities $Q = \{0, \frac{1}{M}, \frac{2}{M}, ..., \frac{M}{M}\} \subset [0, 1]$ with M indivisible shares. Let $\hat{\tau} \equiv \min\{x_0 | x_0 \geq \frac{c}{\Delta v} - \alpha_0, x_0 + \alpha_0 \in \Delta\}$. Equilibrium strategies are given as follows. The distinguished player submits a buy order to buy $M\hat{\tau}$ shares using \underline{v} as a price limit, i.e. submits a limit buy order $\beta = (\hat{\tau}, \underline{v})$. Furthermore, $M\hat{\tau} + 1$ equilibrium sellers submit sell limit orders $\sigma = (\frac{1}{M}, \underline{v})$. One outside investor submits a market buy order for one share, i.e. an order with $\beta = (\frac{1}{M}, \infty)$ if $(1 - \frac{1}{M}) \cdot \Delta v \geq c$. All other investors do not submit any orders.

Clearly, $p^* = \underline{v}$ because trade does not occur at any other price. Further, note that the distinguished player exerts effort if his ex post ownership stake is $\alpha_0 + \hat{\tau}$ but not for smaller ex post ownership stakes by construction of $\hat{\tau}$. Moreover, the equilibrium buyer is an equilibrium winner if $\Delta v \geq \gamma$.

To show that this is indeed an equilibrium consider potential deviations.

- 1. Sellers cannot benefit from not selling since they are pivotal and would trigger the distinguished player to cut back on his effort. As a result, unsold shares would be worth \underline{v} which yields equilibrium utility or eventually even less if they own shares they do not sell in equilibrium.
- 2. A similar argument applies to an outside investor who wants to increase buy orders. To buy more is only feasible if the distinguished player withdraws effort afterwards; hence more buying is not a beneficial deviation.
- 3. The distinguished player cannot benefit from deviating. Increasing his ex post ownership stake ω_0 is not feasible since there are no other sell offers on the market and buying at a higher price is not a beneficial deviation.

Reducing his ownership stake is not beneficial either since exerting effort is beneficial and hence $(\alpha_0 + x_0) \cdot \bar{v} - c \ge (\alpha_0 + x_0) \cdot \underline{v} \ge (\alpha_0 + x_0 - \epsilon) \cdot \underline{v} - (x_0 + \epsilon) \cdot p^*$ if $\epsilon \in M$ and if $\epsilon \le \hat{\tau}$.

5 Trade Equilibrium Characterization

Having shown existence for some specific but relevant trading rules, we continue with the more general results, the characterization of excess returns equilibria.

Theorem 3 Let a^* be an excess returns equilibrium. Then, for any market mechanism μ the following properties are satisfied.

- 1. In equilibrium a^* the distinguished player exerts effort $e^* > 0$.
- 2. In equilibrium a^* the distinguished player i = 0 is not a seller $\omega_0 \ge \alpha_0$.
- 3. Each investor is pivotal in the sense that selling less than specified or buying more than specified by equilibrium strategies triggers the distinguished player i = 0 to sell more or buy less and to reduce effort subsequently or it triggers a price increase.
- 4. The distinguished player submits a buy or sell order.
- 5. In an excess returns equilibrium with an equilibrium winner traditional no arbitrage does not hold while generalized no arbitrage holds by definition.

Proof to be found on page 41.

No distinguished player. This formulation of the model contains the special case $\Delta v = 0$ with no distinguished player. The following proposition shows that models without distinguished players have no excess returns equilibria and therefore are not robust with respect to the introduction of arbitrarily small distinguished players $\Delta v > 0$ if excess returns equilibria exist.

Proposition 1 For a model without a distinguished player $\Delta v = 0$ excess returns equilibria do not exist and therefore traditional no arbitrage is always satisfied.

Proof to be found on page 41.

Our interpretation at this point is, first, that the presence of a distinguished player should turn our attention to both, true value equilibria and to excess returns equilibria. Second, market models assuming that distinguished players cannot trade and influence the firm's value at the same time are not robust with respect to the introduction of a distinguished player.

True Value Equilibria. We have seen in Theorem 1 that true value equilibria do not exist in standard call auctions. Nevertheless, to improve our understanding we are interested in a positive result and ask the question under which conditions they can exist in rational markets and if they exist, how they look like. It turns out that a positive bid ask spread is another obstacle for the existence of true value equilibria; at least in a standard world of price taking agents.

Proposition 2 Consider market game Γ_{μ} defined by market mechanism μ , bid ask spread γ , and initial ownership α .

- 1. For strictly positive bid ask spread $\gamma > 0$ high true value equilibria do not exist in which no investor $k \in \{1, ..., N\}$ is pivotal in the sense that selling less or buying more than specified by equilibrium strategies triggers the distinguished player i = 0to sell more or buy less and to reduce effort subsequently.
- 2. For strictly positive bid ask spread $\gamma > 0$, low true value equilibria do not exist, and true value equilibria do not exist if effort is discrete and $\alpha_0 < c/\Delta v$.
- 3. Any true value equilibrium satisfies traditional no arbitrage.
- 4. Assume from here bid ask spread $\gamma = 0$ and effort to be discrete. For $\alpha_0 \Delta v > c$ a high true value equilibrium exists if the market mechanism μ maximizes the trade volume and if either quantities are continuous or the number of shares M is sufficiently large i.e. $M > \frac{\Delta v}{\alpha_0 \cdot \Delta v c}$. If conversely $\alpha_0 \Delta v < c$ and $\underline{v} > 0$ a low true value equilibria exists if the market mechanism μ maximizes the trade volume and if either quantities are continuous or if M is sufficiently large.

Proof to be found on page 41.

Clearly, to be an equilibrium buyer for strictly positive bid ask spread $\gamma > 0$ in the high true value equilibrium is undesirable since equilibrium buyers must pay $\bar{v} + \gamma$ for something that is known to be worth \bar{v} in equilibrium. The only way to keep buyers happy in a true value equilibrium is pivotalness, i.e. the off equilibrium threat of the distinguished player to decrease their payoff even further.

Traditionally noise traders had to be introduced to initiate trade as long as transaction cost was strictly positive. This observation dates back to the work on no-trade theorems, as for example in Milgrom and Stokey (1982). Our proposition 2 shows that the presence of a distinguished player adds another explanation to initiate trade without noise. More intriguingly, proposition 2 rules out trade equilibria with bid ask spread $\gamma > 0$ unless some traders are pivotal which means that they are 'forced' to sell by the off-equilibrium threat of distinguished player not to exert effort otherwise. Since in true value equilibria the distinguished player has no reason to enforce trade at the true price without noise traders no trade seems to be as realistic.

Proposition 2 together with Theorem 1 indicate that true value equilibria may exist, but only under very restrictive assumptions. True value equilibria only exist for certain combinations of bid-ask spread, ownership structures, and effort specifications. This is a lack of robustness that does not occur for excess returns equilibria. They exist, for example, under continuous effort and zero bid-ask spreads, or for positive bid ask spread, discrete effort and $\alpha_0 < c/\Delta v$ whereas true value equilibria do not exist in these settings. This is not to say that excess returns equilibria may not be plagued by other robustness issues, for example the implied high rationality of outside investors. Rather, both, excess returns equilibria and true value equilibria may be relevant, and we think it is an important task to understand excess returns equilibria.

In the following section we undertake a first step and analyze the robustness of excess returns equilibria when irrational noise traders are introduced and the number of outside investors is allowed to be arbitrarily large.

6 Noise and Small Price Takers

We have seen in proposition 1 that excess returns equilibria do not exist without a distinguished player. In this section we show that other seemingly salient ingredients of the previous fully rational model in fact are not crucial. In particular, the existence of an excess returns equilibrium in a model with a distinguished player does neither rely on traders' rationality and pivotalness nor on their strategic influence on the price. Mathematically, the least form of strategic influence on prices by rational traders is modelled by a continuum of traders. It turns out that if there are irrational noise traders an excess returns equilibrium can be established even in an environment with a continuum of traders. The logic is the following. By some traders' irrationality the price and

final allocation become random variables. Moreover, irrationality raises the chances for rational investors to gain by buying from other traders than the distinguished player. This notably includes the possibility that in this equilibrium small rational investors can behave as price takers and still gain in expectation. These traders face the following tradeoff. To raise the buying price limit above the equilibrium trade price raises the chances to buy for an attractive price below the equilibrium value. However, the downside is that this strategy makes it also more likely to trade against the distinguished player at some higher price that in turn happens to exceed equilibrium value.

Specifications. The set of investors in this section is given by

$$i \in I = [0, 1] = DP \cup RI \cup NT = \{0\} \cup (0, 1) \cup \{1\}$$

consisting of three types of investors. As before the distinguished player is i = 0 and small²² rational outside investors are $i \in (0, 1)$. The distinguished player initially owns proportion $\alpha_0 \geq 0$ of shares and rational outside investors together own $\alpha_r < 1-\alpha_0$ shares. In this section moreover we suppose the presence of irrational noise traders trading for exogenous reasons.²³ Since only their aggregated behavior matters for rational investors they are treated from here as if they were a single irrational investor i = 1. These noise traders initially own together the remaining $\alpha_1 = 1 - \alpha_0 - \alpha_r$ shares.²⁴

The distinguished player is assumed to be strictly wealth constrained and faces again a binary effort choice $e \in \{0, 1\}$. To make sure that best responses are well defined we further assume that every rational investor is budget constrained with a finite budget. The aggregated budget constraint across all rational investors is non-binding and larger than \bar{v} meaning that jointly outside investors can afford to buy the entire firm even at the highest reasonable price.

Prices are discrete $P = P_{\delta} := \{..., \underline{v}, \underline{v} + \delta, ..., \overline{v}, ...\}$ with tick size δ and quantities Q = [0, 1] are continuous.

Noise. Suppose noise traders only submit market orders and only the excess demand correspondence of noise traders denoted by \tilde{Z}_1 matters for the rest of the market²⁵. We further suppose that \tilde{Z}_1 is a random variable with support $[-\alpha_{\theta}, b] \subset \mathbb{R}$ where $-\alpha_{\theta} < 0 < b \leq \alpha_{\rho}$. The assumption $b \leq \alpha_{\rho}$ means that the event $\tilde{Z}_1 > \alpha_{\rho}$ that noise traders want to buy more than rational investors own has probability 0. We introduce this assumption to

 $^{^{22}\}mathrm{In}$ the finance literature continuum traders are often called "atomistic".

²³The presence of noise traders is often motivated by exogenous liquidity shocks.

²⁴More formally, initial ownership structure $\alpha \in \Delta$ in this section is a (probability) measure with $\int_{I} \alpha_{i} di = 1$. Among rational investors we allow here for a finite set of block owners, in particular the distinguished player i = 0 typically is a block owner with $\alpha_{0} > 0$. The big majority of investors of Lebesgue measure 1 are small investors who individually own 0 and only jointly own a strictly positive fracion of shares.

²⁵This is the case anyway if the market mechanism executes market orders against each other.

make sure that existence of excess returns equilibria is not driven by the specification of noise. The distribution function F is allowed to be discontinuous. $F_{-}(z) := \Pr(\tilde{Z}_{1} < z)$ and $F_{+}(z) := \Pr(\tilde{Z}_{1} \leq z)$ of random variable \tilde{Z}_{1} then satisfies $F_{-}(-\alpha_{\theta}) = 0$ and $F_{+}(b) =$ 1. In particular, in contrast to the previous sections we suppose non-degenerate noise, i.e. $\Pr(\tilde{Z}_{1} = 0) = F_{+}(0) - F_{-}(0) = 0$ and $F(0) \equiv F_{+}(0) = F_{-}(0) \in (0, 1)$ meaning that the events $\Pr(\tilde{Z}_{1} > 0) = 1 - F(0) > 0$ and $\Pr(\tilde{Z}_{1} < 0) = F(0) > 0$ have strictly positive probability. Furthermore, we will say that noise increases from low noise l to high noise h represented by distributions F_{l} and F_{h} if

- 1. $\operatorname{Pr}_h(\tilde{Z}_1 > \tilde{x}) \ge \operatorname{Pr}_l(\tilde{Z}_1 > \tilde{x})$ for all $\tilde{x} \in (0, b]$ and $\operatorname{Pr}_h(\tilde{Z}_1 > \tilde{x}) > \operatorname{Pr}_l(\tilde{Z}_1 > \tilde{x})$ for some $\tilde{x} \in (0, b]$
- 2. $\operatorname{Pr}_{h}(\tilde{Z}_{1} < \tilde{x}) \geq \operatorname{Pr}_{l}(\tilde{Z}_{1} < \tilde{x})$ for all $\tilde{x} \in [-\alpha_{\theta}, 0)$ and $\operatorname{Pr}_{h}(\tilde{Z}_{1} < \tilde{x}) > \operatorname{Pr}_{l}(\tilde{Z}_{1} < \tilde{x})$ for some $\tilde{x} \in [-\alpha_{\theta}, 0)$.

Theorem 4 There exist initial ownership structures α and effort cost c such that for a sufficiently small tick size $\delta > 0$

- (i) an excess returns equilibrium exists under the NYSE market microstructure μ_N for any non-degenerate distribution F,
- (ii) an excess returns equilibrium exists under the absolute size priority market microstructure μ_S for any non-degenerate distribution F,
- (iii) and an excess returns equilibrium exists under the Tokyo market microstructure μ_T for some non-degenerate distribution F.
- (iv) Suppose noise increases from low noise l to high noise h and the market microstructure is either NYSE μ_N or absolute size priority μ_S . Then, there always exist α_0, c, a^* such that a^* is an excess returns equilibrium under high noise h while a^* is not an equilibrium under low noise l.
- (v) Without noise, an excess returns equilibrium cannot exist with a continuum of traders under any call auction mechanism.
- (vi) Without a distinguished player ($\bar{v} = \underline{v}$) excess returns equilibria do not exist under the NYSE, Tokyo or absolute size priority market microstructure under any nondegenerate distribution F.

Proof to be found on page 42.

It should be noted that parameters can easily be specified such that excess returns can be substantial. Furthermore, for any noise, excess returns equilibria exist for a whole range of cost parameters and for every cost parameter it holds for a whole range of ownership structures.

The following line of arguments provides the main intuition of the proof and shows why investors have no incentive to bid up the share price. In our excess returns equilibria with noise, the distinguished player sells his shares with strictly positive probability whenever $p \ge p_0$ and shares are overvalued at these high prices which implies that $p_0 \in (\underline{v}, \overline{v})$. However, for any price $p \leq p_0 - \delta$, the distinguished player does not sell his shares. As a result, shares are undervalued if the resulting price is $p \leq p_0 - \delta$. Hence, rational investors want to sell if $p \ge p_0$ and to buy if $p \le p_0 - \delta$. Hence, the value of the firm is price dependent and investors are always rationed: There is excess demand if $p \leq p_0 - \delta$ and excess supply if $p \ge p_0$. Principally, rational investors can overcome the rationing by increasing their buy limits. The downside from this strategy is that they always buy shares, which are sometimes undervalued and sometimes overvalued. Since the rationing factor is determined by comparing noise against rational investors, an increase of noise facilitates the existence of excess returns equilibria for two reasons. First, liquidity is increased which makes it more likely that the distinguished player can sell his shares on the market. Secondly, the rationing problem is reduced which implies that rational investors have a smaller incentive to increase their limit price used in their buy orders.

It is interesting to compare the noisy environment with our earlier analysis. First of all and most important, our voluntary trade property of market mechanisms no longer holds. Noise traders are irrational and are forced to trade for exogenous reasons.

Since the voluntary trade property does not hold for all investors our characterization results of section 5 are not valid any more. In particular, *pivotalness* ceases to hold. In the voluntary trade framework, pivotalness was the key to produce liquidity. Investors had an incentive to trade in equilibrium to guarantee that the distinguished player does not sell his shares. In the present context, the role of providing liquidity is taken by the noise traders. Therefore, excess returns equilibria exist even if investors are small price takers who are not pivotal.

7 Related Literature

The paper relates to empirical and theoretical contributions, in particular those jointly addressing corporate governance and asset pricing. We will first discuss the empirical literature and argue that there is evidence for i) excess returns equilibria, ii) the existence of distinguished players and iii) non-atomistic, pivotal investors with price impact. Second, we discuss related theoretical contributions.

In order to support the excess returns equilibrium phenomenon formulated by this theory, it is first necessary to identify a distinguished player. As potential candidates, von Lilienfeld-Toal and Rünzi (2007) investigate owner-CEOs. They show that the presence of an owner-CEO is not priced. They consider the universe of S&P 1500 (S&P 500) firms from 1996-2005 (1994-2005) and find that a portfolio consisting of S&P 1500 firms in which the CEO owns more than 10% of shares of the company produces statistically

significant annualized abnormal returns of approximately 12%. These results carry over to several robustness checks including a formulation of industry adjusted returns and splitting the sample into boom and burst years. Moreover, von Lilienfeld-Toal and Rünzi (2007) investigate to which extent excess returns differ if effort of the CEO is important. To do so, they sort firms with regard to managerial discretion. They find that excess returns tend to be more important for firms in industries where the CEO has a strong influence on firm performance, younger firms, and growth firms. Other potential candidates for distinguished players are founder-CEOs. Fahlenbrach (2007) finds that founder-CEO firms outperform the market by approximately 10% and these results are again robust to various specifications. While these results are consistent with excess returns equilibria, they are inconsistent with true value equilibria. Hence, true value equilibria may not only fail to exist in theory, empirical evidence also suggests that excess returns equilibria are more relevant, provided good candidates for distinguished players are found.

Aforementioned papers also document the empirical importance of distinguished players and value increasing shareholders for listed US firms. For example, within the S&P 1500 firm universe, according to von Lilienfeld-Toal and Rünzi (2007) more than 10% of firms have an officer who owns more than 10% of outstanding stocks. Fahlenbrach (2007) reports that founder-CEOs are present in 11% of the largest US firms (founders hold on average 11% of shares of a firm). A similar emphasis is put forward in the recent paper by Holderness (2006) with the title "The Myth of Diffuse Ownership in the United States". In a representative sample of 375 US firms he reports that 96% of US firms have at least one blockholder who owns more than 5% of shares of the firm. Average ownership of all blockholders, directors, and officers is 43% (median 43%), average ownership of the largest shareholder is 26% (median 17%), and average ownership of officers and directors is 24% (median 17%). The latter finding is consistent with results in Fahlenbrach and Stulz (2007) who look at the much larger universe of US firms covered in the compact disclosure discs. They analyze 27, 636 firm years from 1988-2003 and report mean ownership of officers and directors to be 22.4% (median 15.8%).

One group of shareholders that are reasonable candidates for non-price taking, pivotal outside investors are institutional investors who control more than 100 million US dollars. It is well documented that trades of this group of investors have an influence on the price. Chan and Lakonishok (1995, p. 1147) argue that "For many institutional investors, however, even a moderately-sized position in a stock may represent a large fraction of the stock's trading volume". They document an average price impact of 1% for buy orders or -.35% for sell orders. Their sample consists of NYSE and AMEX trades of 37 large investment management firms from July 1986 until the end of 1988. Noteworthy, the trades of these 37 institutional investors accounted for approximately 5% of trading volume on NYSE and AMEX in this time period.

Apart from the importance of trading volume of institutional investors, it is also known that the ownership of institutional investors is economically significant. Gompers and Metrick (2001) consider the holdings of institutional investors from 1980-1996. Shareholdings of institutions is increasing over time and in December 1996, the last quarter of their sample, institutional investors hold more than 50% of the market capitalization of US firms. We interpret these observations that there are only a few important institutional investors as supportive for the assumption that outside investors can act strategically rather than as pure price takers. In December 1996, there are only 1303 institutions. In particular, the largest 100 institutions hold approximately *one third* (37.1%) of the entire market capitalization and the largest 10 institutions hold 14.6% of market capitalization of all US firms.

Most of the theoretical literature about large shareholders and trading games are almost exclusively focussed on what we call true value equilibria. Prominent examples include Shleifer and Vishny (1986), Admati, Pfleiderer and Zechner (1994), Maug (1998), DeMarzo and Urosevic (2006), Kahn and Whinton (1998), or Magill and Quinzii (2002). All these papers study a large and value increasing shareholder who may increase a firm's value while increasing a firm's value causes private effort costs. Some of them are more general in other important respects (asymmetric information, dynamic framework, ...) while our paper is more general with respect to the market microstructure and is the only one to investigate and characterize excess returns equilibria.

With respect to excess returns equilibria closest to our paper is von Lilienfeld-Toal (2005), who identifies an excess returns equilibrium for a very specific market microstructure and then turns attention to the empirical implications of excess returns. In particular, it argues that excess returns equilibria are consistent with i) negative abnormal returns around unlock days and ii) positive abnormal returns for firms with a distinguished player. In contrast to our present theory it contains no general theory, no characterization, no statements on non-existence of the true-value equilibrium or on irrational traders.

Note that excess returns equilibria may occur in the model of Bolton and von Thadden (1993) which also is concerned with corporate control issues. It does not focus, however, on (asset) pricing implications of excess returns equilibria and in particular does not relate excess returns to no-arbitrage in asset pricing. Rather, they are mainly interested in the question when blocks of shares remain, vanish or are newly created. The reason as to why excess returns equilibria may exist in the model of Bolton and von Thadden (1993) is similar to our notion of pivotalness. A related explanation of takeovers is given by Bagnoli and Lipman (1985) or Holmstrom and Nalebuff (1992). The latter papers analyze potential solutions to the free rider problem first mentioned by Grossman and Hart (1980).²⁶

All above mentioned papers derive the results using a particular market microstruc-

²⁶In fact, in a sense our model could be interpreted as a generalization of Bagnoli and Lipman (1985) and Holmstrom and Nalebuff (1992) if the distinguished player's value-enhancing capability only unfolds for $\alpha_0 \geq \frac{1}{2}$, the strategy space of the distinguished player is limited to a takeover bid, and other shareholders can only submit sell orders.

ture. We are not aware of other papers deriving characterization results that are valid for a broad class of market micro-structures. Conversely, our assumptions on complete symmetric information in a static market game are clearly more specific in other respects than a large number of the above mentioned contributions.

Our setting can be viewed as double sided auctions with strategic trading and the paper relates to this branch of market microstructure theory. Papers falling within our framework are for example Kyle (1985), Kyle (1989), Rochet and Vila (1994), or Reny and Perry (2006). While their exact specification of price setting and quantity allocation rules is within our class of market mechanisms, the economic environment we are interested in is distinct as compared to these papers. Moreover, market microstructure theory is also interested in the price impact of individual trades which is aptly pointed out by O'Hara (2003): "... asset pricing ignores the central fact that market microstructure focuses on: Asset prices evolve in markets".

Our paper is also related to the literature on no trade theorems, for example Milgrom and Stokey (1982) or Tirole (1982). The driving force behind no trade theorems is the fact that there are no gains from trade or negative gains from trade in the presence of transaction costs. In the class of models we are interested in, gains from trade are zero for true value equilibria and consequently, true value equilibria in the traditional sense fail to exist for positive bid ask spreads. In excess returns equilibria, in contrast, gains from trade are no longer zero sum since the owner manager's threat to sell is viable and trade at a low price prevents the owner manager from selling.

Finally, the paper relates to the literature on agency problems as in Holmstrom (1979) or Grossman and Hart (1983). In particular, models with bilateral contracting and non-exclusive contracts are concerned with externalities among trading partners. Examples are Bisin and Guaitoli (2004), Bizer and DeMarzo (2004), Kahn and Mookherjee (1998) and Segal and Whinston (2004). In these papers, a distinguished player can write contracts with many players while in our model, the distinguished player can anonymously trade with many outside investors.

8 Extensions

It is straightforward to set up a similar theory with more than one distinguished player j = 0, ..., M each of them characterized by his stake, effort cost and potential value $\alpha_j, c_j, \Delta v_j$. In an efficient equilibrium those for which $c_j < \Delta v_j$ should exert effort. Clearly, since distinguished players can behave as regular investors as well the presence of a single distinguished player is sufficient to raise the possibility of excess returns equilibria. For empirical testing we expect that the significance of the most important distinguished players since triggering this player to sell causes the biggest potential payoff losses. If the market price offers the right incentives for this most important distinguished player it should as

well do for minor distinguished players.

We have not discussed the potential role of market makers in our model even though our market microstructure allows for an analysis of market makers. Analyzing the role of market makers can be done as follows. Suppose agent N + 1 is the market maker. Moreover, agent N + 1 submits buy and sell orders $(1, \infty)$ and $(1, -\infty)$. In other words, the agent N + 1 submits to buy and sell the entire firm. Hence, any market imbalance can be bought by agent N + 1 or sold to agent N + 1 without violating our "voluntary trade" assumption. Moreover, the market maker is hardwired such that he maximizes u_{N+1} , potentially subject to institutional rules such as market clearing, i.e. offering a clean up price. The resulting allocations and prices will be the same as in a market mechanism in which all agents i = 0, 1, ..., N submit their shares, a market maker observes these orders and sets a price to maximize his utility, subject to obeying institutional rules.

Since our model in principle exhibits multiple equilibria, empirical observations might help to judge which equilibria are most relevant in reality. Building on this observation, the following questions could be addressed in future empirical research: How general is the occurrence of excess returns equilibria for smaller firms than S&P1500? How do abnormal returns increase or decrease with firm size and with the initial stake of the distinguished player? Can we detect the importance and magnitude of private costs of control from these excess returns equilibria? Are there any differences in trading volume for firms with and without a distinguished player, especially if we control for transaction costs? Moreover, it might be possible to test more general aspects of game theory using stock market data and the methodology of games in aggregated form developed in Blonski (2005) which allows robust predictions on the structure of large semianonymous games without specific knowledge about individual preferences. The numerous implications of excess returns equilibria promise further interesting combinations of theoretical and empirical work on incentives, game theory and asset pricing.

Our model is a static model which does not incorporate any dynamic aspects of real world trading. Therefore, it is important to see how our model fares in a dynamic version. Finally, as emphasized in the introduction information is typically asymmetric if distinguished players trade in stock markets. It is an important, interesting and challenging project to study the interplay between the structure of excess returns equilibria with informational asymmetries. As the notion of no-arbitrage equilibrium in traditional capital market valuation theory generalizes from the complete information case to the case of information asymmetries we conjecture that the notion of generalized no-arbitrage as formulated in this article generalizes as well to the case with informational asymmetries.

9 Conclusions

To summarize our main findings, we show that shares of a firm with a distinguished player can often not be priced correctly. True value equilibria do not exist under positive trading costs or if effort is continuous and trading anonymous. In contrast, excess returns equilibria exist in both, a fully rational world and a world with noise and a continuum of traders. We characterize excess returns equilibria for a general class of market mechanisms. It turns out that the existence of a distinguished player is necessary for excess returns equilibria to exist.

Our theory is general in the sense that it contains the benchmark case of a frictionless efficient market without distinguished player and without transaction costs and with the usual true value equilibria as a special case. In this sense our results show that pricing predictions of models with frictionless markets are not robust with respect to the introduction of an arbitrarily small²⁷ distinguished player.

The main intuition behind the existence of excess returns equilibria is as follows. Whenever share prices of a firm exceed a certain threshold, the distinguished player prefers to sell his shares – or does not want to buy shares – and reduce effort subsequently. As a result, shares are traded below this threshold price. Due to the private effort costs, this threshold price is below the equilibrium value.

Now, trade can occur for two reasons. Equilibrium sellers – selling shares below the equilibrium value – can be pivotal and highly rational. They then know that not selling shares will trigger the distinguished player to sell shares instead, cut back on the costly effort and reduce firm value. This renders everyone worse off, including deviating equilibrium sellers. In the absence of pivotal traders, noise traders may fill the liquidity gap. Noise traders may sell for exogenous and stochastic reasons. Then, the equilibrium share price is stochastic and the distinguished player has an incentive to sell at the high realizations of the share price but not at the low realizations of the share price. Rational buyers may then prefer not to buy at high share prices but only buy at low share prices. As a consequence, they are not increasing demand – by submitting bids with higher buy limits – and hence they are not increasing the share price even though on average, shares are traded below the equilibrium value.

Additional research questions arise naturally. For example, what happens in a dynamic formulation or under asymmetric information or with risk aversion? While the economic intuition behind our results is quite strong, it is also apparent that a rigorous formulation of such questions is not straightforward. All our proofs turn out to be involved and full of details.

Since there are two different explanations for trade to occur in an excess returns equilibrium (fully rational, pivotal players or irrational noise traders), it would be interesting to empirically account for the importance of each explanation. Even though there exists some evidence for excess returns equilibria (von Lilienfeld-Toal and Rünzi (2007) and Fahlenbrach (2007)), it is a worthwhile task to identify excess returns equilibria in other circumstances or conversely to identify circumstances where excess returns do not exist.

 $^{^{27}\}mathrm{The}$ size of the distinguished player is defined by his maximal contribution to the company's fundamental value.

Investigating different aspects and puzzles of asset pricing, both in theoretical and empirical work also promises to be fruitful. The importance to carefully investigate asset pricing phenomena in light of our theory becomes clear when it comes to judging the empirically observed excess returns. Observing abnormal returns due to a certain investment strategy, as documented by von Lilienfeld-Toal and Rünzi (2007), need not be a sign of irrational behavior but might be the result of excess returns equilibria and highly rational behavior.

10 Appendix A: Orders and Strategies

Orders. Demand and supply correspondences can be composed by sets of orders. The following description of the relevant market actions denoted as *orders* is chosen to model real world market mechanisms as closely as possible. First, denote by $B = Q \times P$ the space of *buy limit orders* with typical element $\beta = (b, p)$ where $p_b = p + \gamma \in P$ denotes the limit price up to which a player is willing to buy any quantity $q \leq b \in Q$. Conversely, for a *sell limit order* $\sigma = (s, p) \in S = Q \times P$ the price p is the minimal price from which the submitting trader is willing to sell q up to quantity s. Buy and sell orders can be interpreted as downward sloping step correspondences. For example, the buy limit order $\beta = (b', p')$ is precisely defined by the correspondence $\beta : P \to Q$ where

$$\beta(p) = \begin{cases} q \in Q | q \leq b' \\ 0 & \text{otherwise} \end{cases} \quad \text{for } p \leq p' + \gamma$$

A trader who submits $\beta = (b, \infty)$ or $\sigma = (s, -\infty)$ is said to submit a market order since a certain quantity is ordered for buy or sell independently of price. Market order correspondences are bounded by vertical lines. Similarly, the graphs of *buy stop orders* denoted by $\chi = (b_{st}, p)$ and *sell stop orders* denoted by $\psi = (s_{st}, p)$ are upward-sloping step correspondences. The interpretation is that any quantity up to b_{st} is bought above price p or quantity s_{st} is sold below price p. We also allow so called *fill or kill orders* or *all or nothing orders* that specify that a certain quantity is to be bought or sold entirely or not at all. A fill or kill order is denoted by $\mathring{\beta} = (b, p), \mathring{\sigma}, \mathring{\chi}, \text{ or } \mathring{\psi}$. More precisely, say for $\mathring{\beta} = (b', p')$, the related correspondence $\mathring{\beta} : P \to Q$ has a non-convex graph and is defined as

$$\mathring{\beta}(p) = \begin{cases} \{0, b'\} & \text{for } p \le p' + \gamma \\ 0 & \text{otherwise} \end{cases}$$

Strategies. A market game strategy a_i of player i = 0, ..., N is a collection of orders

$$\begin{array}{lll} a_i & = & \{(\beta_i^1, \beta_i^2...), (\mathring{\beta}_i^1, \mathring{\beta}_i^2, ...), (\sigma_i^1, \sigma_i^2, ...), (\mathring{\sigma}_i^1, \mathring{\sigma}_i^2, ...), \\ & & (\chi_i^1, \chi_i^2...), (\mathring{\chi}_i^1, \mathring{\chi}_i^2, ...), (\psi_i^1, \psi_i^2, ...), (\mathring{\psi}_i^1, \mathring{\psi}_i^2, ...)\}. \end{array}$$

Denote by A_i the corresponding *strategy space* of player *i* and by $A = A_0 \times \cdots \times A_N$ the strategy profiles. Adding up buy and sell orders for some player *i* yields the individual excess demand correspondence

$$Z_{i}(p) = z_{i}(p) + \mathring{z}_{i}(p) \text{ composed by}$$

$$z_{i}(p) = \sum_{\beta,\sigma,\chi,\psi\in a_{i}} \beta(p) + \chi(p) - \sigma(p) - \psi(p) \text{ and}$$

$$\mathring{z}_{i}(p) = \sum_{\mathring{\beta},\check{\sigma},\mathring{\chi},\check{\psi}\in a_{i}} \mathring{\beta}(p) + \mathring{\chi}(p) - \mathring{\sigma}(p) - \mathring{\psi}(p)$$

adding up buy and sell orders²⁸ of player i. A market game strategy can be decomposed into buy orders and sell orders. Denote by

$$D_i(p) = d_i(p) + \mathring{d}_i(p) = \sum_{\beta \in a_i} \beta(p) + \sum_{\chi \in a_i} \chi(p) + \sum_{\mathring{\beta} \in a_i} \mathring{\beta}(p) + \sum_{\mathring{\chi} \in a_i} \mathring{\chi}(p)$$

$$S_i(p) = s_i(p) + \mathring{s}_i(p) = \sum_{\sigma \in a_i} \sigma(p) + \sum_{\psi \in a_i} \psi(p) + \sum_{\mathring{\sigma} \in a_i} \mathring{\sigma}(p) + \sum_{\mathring{\psi} \in a_i} \mathring{\psi}(p)$$

player *i*'s individual demand and supply correspondences given as quantities player *i* is willing to buy or to sell at a given price *p*. In particular $d_i(p)$ and $s_i(p)$ specify individual demand and supply excluding fill or kill orders.

11 Appendix B: Stochastic Market Mechanisms

Stochastic trade equilibria. Real world market mechanisms often are specified by a list of rules with decreasing order of priority. Sometimes there remains some ambiguity with respect to equilibrium price or allocation if all rules are satisfied by more than one price and/or set of executed orders such that a random choice may be implemented. In this section we consider risk neutral investors facing a stochastic market mechanism.

A stochastic ownership structure $\tilde{\xi} \in \tilde{\Delta}$ is an element of the space of probability measures $\tilde{\Delta}$ on simplex Δ . Accordingly, define stochastic market prices $\tilde{p} \in \tilde{P}$ and stochastic trade vectors $(\tilde{x}, \tilde{y}) \in \tilde{\Phi}$.

Definition 3 For any initial ownership $\alpha \in \Delta$ and strategy profile $a \in A$ a stochastic market mechanism $\tilde{\mu}$ with bid ask spread γ is a mapping

$$\tilde{\mu} : \Delta \times A \to \tilde{P} \times \tilde{\Phi}$$

where for initial ownership α and strategy profile a the market mechanism $\tilde{\mu}(\alpha, a) = (\tilde{p}, \tilde{x}, \tilde{y})$ picks a stochastic sell price \tilde{p} (and buy price $\tilde{p} + \gamma$) and for any player trade is voluntary. This means that only submitted orders can be executed, i.e. for any state of nature $(\tilde{x}_i, \tilde{y}_i) \in D_i(p) \times S_i(p)$ and therefore $\tilde{\omega}_i - \alpha_i \in Z_i(p)$. Again, the stochastic trading volume is

$$\tilde{\tau}_{\mu} : \Delta \times A \to [0, 1]$$

and $\tilde{\tau}_{\tilde{\mu}}(\alpha, a) = \sum_{i=0}^{N} \tilde{x}_i(\alpha, \gamma, a).$

The distinguished player picks his effort decision e after the stochastic market game is over and the realizations of all random variables are known. Denote by \tilde{e} the random effort decision induced by the realization of (\tilde{x}, \tilde{y}) which determines the final stake of the distinguished player. Similar as before, a stochastic market mechanism $\tilde{\mu}$ together with an initial ownership α and bid ask spread γ induces a stochastic market game Γ_{μ} with strategy space A and risk neutral payoff functions given by

$$u_0(a) = E\left[\tilde{\omega}_0 \underline{v} + \tilde{e}\left(\tilde{\omega}_0 \Delta v - c\right) - \left(\tilde{p} + \gamma\right) \tilde{x}_0 + \tilde{p} \tilde{y}_0\right] \text{ and} u_i(a) = E\left[\tilde{\omega}_i \left(\underline{v} + \tilde{e} \Delta v\right) - \left(\tilde{p} + \gamma\right) \tilde{x}_i + \tilde{p} \tilde{y}_i\right] \text{ for } i = 1, \dots N$$

where E means expectation value.

²⁸It is necessary to differentiate in our notation the cases including and excluding fill or kill orders since in most real world market mechanisms kill or fill orders are treated differently. For example, they are not written in the order book and thereby have no direct influence on the market price.

Stochastic true value and excess returns equilibria. A stochastic true value equilibrium is an equilibrium where $E(p^* | \alpha \neq \omega) = E(\tilde{v} | \alpha \neq \omega)$ and a stochastic excess returns equilibrium is defined as an equilibrium where $E(p^* | \alpha \neq \omega) < E(\tilde{v} | \alpha \neq \omega)$.

12 Appendix C: Proofs

PROOF (OF THEOREM 1, PAGE 18) (I) non-existence, (II) existence.

(I) Suppose a^* were a true value equilibrium with $Ev(\omega_0^{\mu}(a^*)) = p^*$. The proof proceeds by showing that the distinguished player i = 0 or some outside investor i > 0 can always improve which is a contradiction to a^* being an equilibrium. The logic of the proof is always as follows: Whenever shares are traded in a true value equilibrium, shares are overvalued from the perspective of the distinguished player and he wants to sell. Whenever shares are priced correctly from the perspective of the distinguished player they are undervalued from the perspective of outside investors.

The proof uses the following auxiliary result that the distinguished player wants to change his expost holdings at the equilibrium price. This is shown by proving the following lemma.

Lemma 1 Consider a candidate true value equilibrium a^* with corresponding ω^*, x^*, y^* and deterministic ω_0^* . Then, strategy a'_0 generating profile $a' = (a'_0, a^*_1, a^*_2, \ldots, a^*_N)$ is a profitable deviation for the distinguished player – i.e. $u_0(a') > u_0(a^*)$ – if the price never changes $p(a') = p(a^*)$ but new ex-post allocation $\omega'_0 \equiv \omega_0^{\mu}(a') \neq \omega_0^*$ occurs with positive probability. Further, if $\omega_0^* \neq \alpha_0$, strategy $a'_0 = 0$ (not trading) is a beneficial deviation.

PROOF Since for a stochastic market mechanism ex-post ownership is a random variable we show the first claim of the lemma for any ex-post realization ω'_0 with $\omega'_0 \equiv \omega^{\mu}_0(a') \neq \omega^*_0$.

For $\omega'_0 \neq \omega^*_0$ we can rewrite $\omega'_0 = \alpha_0 + x^*_0 + \epsilon_x - y^*_0 - \epsilon_y$ with $\epsilon_x \neq \epsilon_y$. Then, expost utility is given as

$$u_{0}(a') = \omega'_{0}Ev(e(\omega'_{0})) - p^{*} \cdot (x_{0}^{*} + \epsilon_{x}) + p^{*} \cdot (y_{0}^{*} + \epsilon_{y}) - c(e(\omega'_{0}))$$

$$> \omega'_{0}Ev(e(\omega_{0}^{*})) - p^{*} \cdot (x_{0}^{*} + \epsilon_{x}) + p^{*} \cdot (y_{0}^{*} + \epsilon_{y}) - c(e(\omega_{0}^{*}))$$

$$= \underbrace{(\omega'_{0} - \epsilon_{x} + \epsilon_{y})}_{\omega_{0}^{*}}Ev(e(\omega_{0}^{*})) - p^{*} \cdot (x_{0}^{*} - y_{0}^{*}) - c(e(\omega_{0}^{*}))$$

$$= u_{0}(a^{*}).$$

for any $\omega'_0 \equiv \omega_0^{\mu}(a') \neq \omega_0^*$. The strict inequality > follows since $e(\omega_0^*) \neq e(\omega'_0) = \operatorname{argmax}_e \omega'_0 E(v(e)) - c(e)$ and the subsequent equations make use of $p^* = Ev(e(\omega_0^*))$ in a true value equilibrium. This implies that the distinguished player improves if $\omega'_0 \equiv \omega_0^{\mu}(a') \neq \omega_0^*$ and is unaffected if $\omega'_0 \equiv \omega_0^{\mu}(a') = \omega_0^*$.

For the special case $\omega_0^* > \alpha_0$, strategy $a'_0 = 0$ (not trading) is a beneficial deviation because then

$$u_{0}(a') = \alpha_{0} Ev (e(\alpha_{0})) - c(e(\alpha_{0}))$$

$$> \alpha_{0} Ev (e(\omega_{0}^{*})) - c(e(\omega_{0}^{*}))$$

$$= [\omega_{0}^{*} - x_{0}^{*} + y_{0}^{*}] Ev (\omega_{0}^{*}) - c(e(\omega_{0}^{*}))$$

$$= \omega_{0}^{*} Ev (\omega_{0}^{*}) - p^{*} \cdot (x_{0}^{*} - y_{0}^{*}) - c(e(\omega_{0}^{*}))$$

$$= u_{0}(a^{*})$$

for the same reasons as in the first part of the lemma.

Now we proceed by considering the following 3 cases. (i) ω_0 is deterministic and the distinguished player always trades, (ii) ω_0 is deterministic and the distinguished player never trades, or (iii) ω_0 is stochastic.

- (i) Suppose first ω_0 is deterministic and $\omega_0^* \neq \alpha_0$. From lemma 1 it follows that $a'_0 = 0$ is a beneficial deviation and hence a^* cannot be a true value equilibrium.
- (ii) Now, suppose ex post ownership ω_0^* is deterministic and $\omega_0^{\mu}(a^*) = \alpha_0$. Consider first the case where $d(p^*) = 0$ where nobody submits limit buy orders. Since we are looking at a trade equilibrium, there must exist fill-or-kill orders that are executed. In this case the distinguished player can mimic one fill-or-kill buy order that is executed with positive probability. By definition of the market mechanisms this will not have a price impact and the deviating fill-or-kill order of the distinguished player will be executed with positive probability. This would constitute a beneficial deviation due to lemma 1. Therefore, in a true value equilibrium holds $d(p^*) > 0$ if ω_0^* is deterministic.

We next show non-negative excess limit-order-demand $d^*(p^*) \ge s^*(p^*)$. If to the contrary $d^*(p^*) < s^*(p^*)$ the distinguished player can improve buy submitting an order $a'_0 = \beta'_0 = (s^*(p^*) - d^*(p^*), p^*)$. As a result, his order will be served and the price will not change. Again, by virtue of lemma 1, a'_0 is a beneficial deviation and hence a^* not an equilibrium.

Next we claim that if $d^*(p^*) \geq s^*(p^*)$ there must exist a buy order $\beta_i^* = (x, p^*)$ with price limit p^* at the equilibrium price for some positive quantity x > 0 which is partially or fully executed with positive probability. Suppose not. Then, $d(p^* + \delta) \geq s(p^*)$. Since $s(p^*) \leq s(p^*+\delta)$ it follows that $\tau(p^*) \leq \tau(p^*+\delta)$. Clearly, $\tau(p^*) < \tau(p^*+\delta)$ is a contradiction of rule 1 from the market mechanism to maximize trade volume. Hence, $\tau(p^*) = \tau(p^* + \delta)$. But then, p^* is picked because the surplus at p^* (the number of unexecuted orders given, price setting rule 2) is (weakly) smaller at p^* than at $p^* + \delta$. The surplus at $p^* + \delta$ is given as $s(p^* + \delta) - d(p^* + \delta)$. This implies that an equilibrium seller can submit a deviating order $a'_i \cup \beta'_i$ with $\beta'_i = (x, p^*)$ where $x > s(p^* + \delta) - d(p^* + \delta)$. This order results in a price increase from p^* to $p^* + \delta$ because now the surplus is greater at p^* . This leads to an improvement for the seller since he can now sell at a higher price (his deviating buy order will not be executed and his equilibrium sell order will be executed due to price priority). This shows that there exists a buy order $\beta_i^* = (x, p^*)$ for some x > 0 which is partially or fully executed with positive probability.

This implies that $\tau^*(p^* + \delta) < \tau^*(p^*)$. Otherwise, no buy order using p^* as a price limit will be executed: All buy orders using $p^* + \delta$ as price limit are executed due to price priority. Now, we claim that the distinguished player can construct a deviating buy order $a'_0 = a_0^* \cup \beta'_0$ with $\beta'_0 = (x, p^*)$ where x is chosen by some other outside investor who is served in the candidate equilibrium with positive probability. As a result, the price will not change (note that $\tau(p^* + \delta | a') < \tau^*(p^* | a')$ continues to hold since the LHS of this equation is not affected by the additional order). Since both orders have the same priority, the distinguished player will now be served with positive probability. Again, lemma 1 implies that the distinguished player improves. Hence, it cannot be an equilibrium that the distinguished player never trades.

(iii) Next, suppose ex post ownership ω_0^* is stochastic. We argue that this cannot be an equilibrium in a sequence of steps. First, we note that there must exist some outside investors whose ex post ownership structure is also stochastic. Then, we show that the ex post ownership of the distinguished player must be a two point distribution (i.e. either he buys a block or he sells a block). Finally, if the distinguished player buys shares, he must be the only buyer (among the set of players with stochastic ownership) who buys and if he sells he must be the only seller (among the set of investors with stochastic ex post ownership). This implies that all agents with stochastic ownership can improve: Buyers only buy if the distinguished player does not buy, hence they buy only overvalued shares. For sellers the same argument apply and they only do not sell if shares are overvalued and they are better off always selling which can be implemented by a deviating strategy.

It is helpful to start with the following lemma.

Lemma 2 Consider any candidate true value equilibrium a^* with stochastic $\omega_0^* \in [\omega_0^{min}, \omega_0^{max}]$ where $\omega_0^{min} < \omega_0^{max}$.

- (a) Then, for any $\omega'_0 \in (\omega_0^{\min}, \omega_0^{\max})$ with $\Pr(\omega_0^* = \omega'_0) > 0$ it is true that $u_0(a^*|\omega_0^* = \omega'_0) < \max\{u_0(a^*|\omega_0^* = \omega_0^{\max}), u_0(a^*|\omega_0^* = \omega_0^{\min})\}.$
- (b) The distinguished player can find strategies a'_0 for which $\lim_{\delta \to 0} u_0(a'_0) \ge u_0(a^*|\omega_0 = \omega_0^{\min})$ if $\omega_0^{\min} \le \alpha_0$ and $\lim_{\delta \to 0} u_0(a'_0) \ge u_0(a^*|\omega_0 = \omega_0^{\min})$ if $\omega_0^{\max} \ge \alpha_0$.
- (c) For small enough tick size δ , ex post ownership of the distinguished player follows a two point distribution, i.e. $\omega_0^* \in \{\omega_0^{min}, \omega_0^{max}\}$ with $\omega_0^{min} \leq \alpha_0 \leq \omega_0^{max}$. Furthermore, the distinguished player is indifferent between ω_0^{min} and ω_0^{max} : $u_0(a^*|\omega_0^* = \omega_0^{max}) = u_0(a^*|\omega_0^* = \omega_0^{min})$.
- (d) It is true that $\omega_j^* = \omega_j^{max} \Rightarrow \omega_0^* = \omega_0^{min}$ for all outside investors $j \neq 0$ who have stochastic expost ownership.

PROOF (a) Suppose first that $p^* \leq E(v(\omega'_0))$, i.e. shares are undervalued if $\omega_0 = \omega'_0$. Then, $u_0(a^*|\omega_0^* = \omega'_0) \leq \omega_0^{max} \cdot E(v(e(\omega'_0))) - p^*[\omega_0^{max} - \alpha_0] - c(e(\omega'_0)) < u_0(a^*|\omega_0^* = \omega_0^{max})$ because $e(\omega'_0) \neq e(\omega_0^{max})$. If $p^* \geq E(v(\omega'_0))$ we get $u_0(a^*|\omega_0^* = \omega'_0) \leq \omega_0^{min} \cdot E(v(e(\omega'_0))) - p^*[\omega_0^{min} - \alpha_0] - c(e(\omega'_0)) < u_0(a^*|\omega_0^* = \omega_0^{min})$ for the same reasons.

(b) Our claim is trivially true if $\omega_0^{min} = \alpha_0$ or $\omega_0^{max} = \alpha_0$. Then, not submitting an order $a'_0 = 0$ yields $u_0(a') = u_0(a^*|\omega_0 = \alpha_0)$.

The distinguished player submits a limit order using p^* as limit price which is randomly executed. Suppose first an order mimicking $\omega_0^{max} > \alpha_0$. Since a limit order is rationed, it follows that $d(p^*) > s(p^*)$ and $\tau^*(p^*) \ge \tau^*(p^*+\delta)$. (using similar arguments as applied in the proof of part(ii) of this proposition.) The distinguished player can now submit a buy order $\beta'_0 = (x, p^* + \delta)$ where x is appropriately chosen to guarantee that $\alpha_0 + x - y^* = \omega_0^{max}$. Note that ex post ownership $\omega'_0 = \omega_0^{max}$ and hence $u_0(a') = \omega_0^{max} \cdot E(v(\omega_0^{max})) - p(a')(\alpha_0 - \omega_0^{max})$. Note also that $p(a') \in \{p^*, p^* + \delta\}$ because $\tau(p^* + \delta|a')$ (weakly) increases and $\tau^*(p^*)$ does not decrease. Hence, either $u_0(a') = \omega_0^{max} \cdot E(v(\omega_0^{max})) - p^*(\alpha_0 - \omega_0^{max})$ and the claim holds for any element of the sequence or $u_0(a') = \omega_0^{max} \cdot E(v(\omega_0^{max})) - (p^* + \delta)(\alpha_0 - \omega_0^{max})$ and for the limit of the sequence $\delta \to 0$ the claim is true.

Next, consider mimicking $(\omega_0^{min} < \alpha_0)$. Then, the distinguished player can now submit a sell order $\sigma'_0 = (y, p^* - \delta)$ with y appropriately chosen to guarantee that $\alpha_0 + x^* - y = \omega_0^{min}$. Similar arguments now imply that the distinguished player now always has $\omega_0 = \omega_0^{min}$ and the price decreases by at most δ .

The distinguished player does not submit a limit order using p^* as limit price which is randomly executed. If a limit order is not rationed, the distinguished player only submits fill or kill orders that are executed stochastically. Now, the distinguished player can replace any set of fill or kill orders that are executed if $\omega_0 = \omega_0^{min}$ (resp. $\omega_0 = \omega_0^{max}$) by limit orders without the fill or kill provision that use p^* as the price limit. If the price does not change $(p(a') = p^*)$, these limit orders will always be executed because they have higher priority than fill or kill orders and the claim is shown. If the price changes due to the proposed deviating strategy, a slightly more complicated strategy must be used to guarantee ω_0^{min} or ω_0^{max} . Consider first $\omega_0^{min} < \alpha_0$. Note first that a change in price implies that $\tau^*(p^*) = \tau^*(p^* - \delta)$ due to trade volume maximizing. Furthermore, $|d^*(p^*) - s^*(p^*)| \leq |d^*(p^* - \delta) - s^*(p^* - \delta)|$. Now, consider any set of fill or kill orders submitted by the distinguished player that leads to ω_0^{min} . This set of fill or kill orders is now replaced by limit orders without the fill or kill provision using p^* as the limit price (for example, $\sigma'_0 = (y, p^*)$ with y chosen appropriately to guarantee ω_0^{min}). Clearly, $\tau(p^*|a') \geq \tau^*(p^*)$. This simple strategy is complemented by a buy order $\beta''_0 = (x'', p^* - \delta)$ with x'' sufficiently large. Since, $\tau(p^*|a') \geq \tau^*(p^*) \geq \tau^*(p^*)$, sufficiently large x'' now guarantees that the surplus is also smaller at p^* under the deviating strategy a'. Then, the price does not change and indeed $u_0(a') = u_0(a * |\omega_0^{min})$.

Next, consider mimicking $\omega_0^{max} > \alpha_0$. Note first if $d^*(p^*) < s^*(p^*)$, the distinguished player can pick any set of executed fill or kill orders executed that leads to ω_0^{max} and replace them by limit orders with the same quantities that use p^* as the limit price. Since this will increase demand $d(p^*|a') > d^*(p^*)$ a price change does not follow. Hence, we are now concerned with the case that $d^*(p^*) \ge s^*(p^*)$. Suppose first that $d^*(p^*) > s^*(p^*)$. Then, the distinguished player can submit buy orders using $p^* + \delta$ as price limit. As a result, the price will increase by at most δ and $\omega_0 \ge \omega_0^{max}$ (the inequality may occur if demand of fill-or-kill orders is reduced at $p^* + \delta$ and the distinguished player has to submit large enough buy orders to guarantee that fill or kill sell orders can be executed.) If $\omega_0(a') > \omega_0^{max}$ note that $p^* < E(v(e(\omega_0^{max})))$ and the distinguished player can buy even more undervalued shares.

Finally, suppose that $d^*(p^*) = s^*(p^*)$ (this can in particular occur if $\tau^*(p^*) = 0$). Since $s^*(p^*) \leq s^*(p^* + \delta)$ it follows that an increase in the share price can only occur due to an increase of demand at p^* if $d^*(p^*) = d^*(p^* + \delta)$. In that case, however, it must be the case that the surplus at $|s^*(p^* + \delta) - d^*(p^* + \delta)| > 0 = |d^*(p^*) - s^*(p^*)|$ due to the last price setting rule since $p^* < \bar{v}$ for a true value equilibrium if ex post ownership $\omega_0 < 1$ with positive probability. From this, it follows $s^*(p^* + \delta) > d^*(p^* + \delta)$ and $\tau(p^* + \delta|a') > \tau(\tilde{p})$ for all $\tilde{p} \notin \{p^*, p^* + \delta\}$. Hence, a buy order using $p^* + \delta$ as a price limit leads to a price increase of at most δ and it will be served since limit orders have higher priority than fill or kill orders.

(c) If the claim is not true, combining (a) and (b) implies that the distinguished player can always find a deviating strategy by mimicking ω_0^{min} or ω_0^{max} . Also, if the distinguished player is not indifferent between ω_0^{min} and ω_0^{max} he can pick a mimicking strategy that approximate the expost ownership that leads to a higher utility. Note that the distinguished player cannot be indifferent between ω_0^{min} and ω_0^{max} if either $\omega_0^{min} > \alpha_0$ or $\alpha_0 > \omega_0^{max}$.

(d) Suppose that the claim $\omega_j^* = \omega_j^{max} \Rightarrow \omega_0^* = \omega_0^{min}$ for all $j \neq 0$ does not hold. Then, there exist a $k \neq 0$ for which $\omega_k^* = \omega_k^{max}$ and $\omega_0^* = \omega_0^{max}$. This implies that outside investor k submits a buy order which is served while $\omega_0^* = \omega_0^{max}$ or outside investor k submits a sell order which is not served while $\omega_0^* = \omega_0^{max}$. Then, the distinguished player can choose the deviating strategy a'_0 which mimicks ω_0^{max} as described under (b) and complement this with a buy order $\beta_0'' = (x'', p')$ where p' is the price used to guarantee ω_0^{max} as derived in (b) and $x'' = \omega_k^{max} - \omega_k^{min}$. This increases utility from the distinguished player because he can now buy more undervalued shares.

This completes the proof of the lemma.

From lemma 2 it follows that outside investors who submit a buy order which leads to stochastic expost ownership can benefit from not buying. Their buy order is only executed if $\omega_0^* = \omega_0^{min}$. However, in a true value equilibrium $p^* = \Pr(\omega_0^* = \omega_0^{min}) \cdot E(v(e(\omega_0^{min}))) + e(v(e(\omega_0^{min}))))$

 $\Pr(\omega_0^* = \omega_0^{max}) \cdot E(v(e(\omega_0^{max})))$. Since $E(v(e(\omega_0^{max}))) > E(v(e(\omega_0^{min})))$ this implies $p^* > E(v(e(\omega_0^{min})))$ and buying outside investors only buy overvalued shares. Hence, not buying constitutes an improvement.

Outside investors who sell stochastically can improve by employing the following deviating strategies. If the distinguished player submits fill or kill orders that are executed randomly, selling outside investors can deviate from submitting fill or kill orders themselves and submit a limit sell order using p^* as the price limit. This limit sell order is accompanied by a large enough (unexecuted) buy order using $p^* - \delta$ as price limit which then guarantees that the price does not decrease. Then, the limit sell order will always be executed before the distinguished player sells his shares. If the distinguished player submits limit orders, outside investors can reduce the price limit of their limit sell orders to $p^* - \delta$. As a result, their order will be executed due to price priority and the price decrease is at most δ . This completes the proof of non-existence of true value equilibria if ex post ownership is stochastic.

Ex post ownership of the distinguished player can neither be deterministic nor stochastic in a true value equilibrium which proves non-existence of a true value equilibrium.

(II) To prove existence of the excess returns equilibrium we construct equilibrium strategies as follows. The distinguished player submits a buy order for one share $a_0^* = \beta_0^* = (1, \hat{p})$ for any $\hat{p} \in]Ev(e(\alpha_0)), \bar{p}[$ with \bar{p} sufficiently close to $Ev(e(\alpha_0))$ and for small enough δ there exist such \hat{p} . It turns out that $\bar{p} \in]Ev(e(\alpha_0)), Ev(e(\alpha_0 + 1/M))[$. Furthermore, one outside investor submits a sell order for one share using \hat{p} as the price limit $a_i^* = \sigma_i^* = (1, \hat{p})$.

Every market mechanism sets $p^* = \hat{p}$ since all other prices lead to zero trade volume and we are looking at an excess returns equilibrium because $\omega_0^* = \alpha_0 + 1/M$, hence $p^* < E(v^*)$ and $\omega \neq \alpha$. Outside investors cannot benefit from increasing demand as this is only feasible if they buy instead of the distinguished player. Then, shares are worth $(Ev(e(\alpha_0))) < \hat{p}$ which is thus not a profitable deviation. The equilibrium seller cannot benefit from not selling since then the share not sold will be worth $(Ev(e(\alpha_0))) < \hat{p}$.

What remains to be shown is that the distinguished player cannot benefit from not trading. This is true if

$$\alpha_0 \cdot E(v(e(\alpha_0))) - c(e(\alpha_0)) < (\alpha_0 + 1/M) \cdot E(v(e(\alpha_0 + 1/M))) - p^* \cdot (1/M) - c(e(\alpha_0 + 1/M))$$

This inequality holds for $p^* = E(v(e(\alpha_0)))$ because $e(\alpha_0) \neq e(\alpha_0 + 1/M)$ and it therefore also holds for a p^* sufficiently close to $E(v(e(\alpha_0)))$.

PROOF (OF THEOREM 2, PART (B), PAGE 18) (i) Amsterdam market mechanism μ_A , (ii) Kyle market mechanism μ_K .

(i) Amsterdam market mechanism μ_A : Specify $\gamma < \frac{c}{\alpha_0}, \delta < \frac{1}{2} \left(\frac{c}{\alpha_0} - \gamma\right)$ and $\alpha_0 \ge c/\Delta v$. To prove existence of an excess returns equilibrium we have to specify a strategy profile a^* such that the price and net trade $(p^*, x, y) = \mu_A(\alpha, a^*)$ induced by the Amsterdam market mechanism satisfy (1) $p^* < \bar{v} - \gamma$, (2) $\omega_0 \Delta v \ge c$ high effort of the distinguished player i = 0, (3) which is a best response for all players i = 0, ..., N and (4) there is an equilibrium winner. Before proving existence under μ_A , note that there exists a unique price $p_0 \in \left[\bar{v} - \frac{c}{\alpha_0}, \bar{v} - \frac{c}{\alpha_0} + \delta\right) \cap P$. The crucial property of this price is that p_0 is the maximal price in P below which the distinguished player prefers not to sell and exerting effort e = 1 to selling all his shares and exerting low effort e = 0 since

$$p_0 < \bar{v} - \frac{c}{\alpha_0} + \delta \Leftrightarrow \alpha_0 \bar{v} - c > \alpha_0 \left(p_0 - \delta \right) \tag{*}$$

We call this property (*). Now, consider strategy profile a^* specified by the following conditions.

- (1) $a_0^* = \mathring{\sigma}_0$ with $\mathring{\sigma}_0(p_0) = (\alpha_0, p_0)$: The distinguished player i = 0 submits a single fill-or-kill order using p_0 as the limit price to sell his entire α_0 stocks.
- (2) Further at least 2 investors $i \neq 0$ submit a fill-or-kill order to buy α_0 stocks using a market order, i.e. for these investors $\mathring{\beta}_i = (\alpha_0, \infty) \in a_i^*$.
- (3) Buying shareholders submit market buy orders (b_i, ∞) with $\sum_{i=1}^{N} b_i = q$, hence the entire market quantity is bought by investors who use market orders. Assume that at least one of the buyers is not a seller at the same time.
- (4) Selling shareholders use limit orders $(s_i, p_0 \delta)$ to sell their shares. Each seller keeps at least a proportion $\frac{\delta}{\Delta v + \delta}$ of his initial stake, i.e. $s_i \leq \frac{\Delta v}{\Delta v + \delta} \alpha_i$. Selling bids sum up to the traded quantity $\sum_{i=1}^{N} s_i (p_0 \delta) = q$.
- (5) Finally, some shareholder $j \neq 0$ submits a limit sell order $\sigma_j(s_j, p_0)$ using a price limit p_0 and a quantity s_j satisfying $\alpha_0 > s_j \ge \max_i s_i$, i.e. between α_0 and the largest equilibrium seller quantity $\max_i s_i$.



Figure 1: Excess returns equilibrium aggregated excess demand correspondence for the Amsterdam rules

First, we observe that applying the Amsterdam market mechanism μ_A to strategy profile a^* implements market price $p^* = p_0 - \delta$ and trade volume $\tau(p^*) = q$. To see this note that the trade volume is maximized and the market clears at $p^* = p_0 - \delta$. For higher prices, there is excess supply, lower prices have trade volume of zero. Inspection of figure 1 yields that the trade volume at $p^* = p_0 - \delta$ is $\tau(p^*) = q$. It remains to show that strategy profile a^* satisfies conditions (1), (2), (3) and (4). Since by specification the bid ask spread satisfies $\gamma < \frac{c}{\alpha_0}$ we have $p_0 < \bar{v} - \frac{c}{\alpha_0} + \delta < \bar{v} - \gamma + \delta$ and hence $p^* = p_0 - \delta < \bar{v} - \gamma$ which shows that condition (1) holds. Since the distinguished player does not trade at p^* and therefore $\omega_0 \Delta v = \alpha_0 \Delta v \ge c$ and condition (2) holds. The presence of an equilibrium buyer who is not a seller guarantees (4). Not surprisingly, the most work is condition (3) to show that nobody can gain by deviating. Instead of tediously checking the universe of all possible deviations we proceed by discussing only optimal deviations. Moreover, any particular deviation will lead to a certain price and allocation given the behavior of all other players. We organize possible deviations first by the resulting price level denoted by \hat{p} . Second, among all deviations implementing price \hat{p} we only consider an optimal deviation for any player i in the sense that no other deviation of this player which implement \hat{p} can yield more.

- 1. $\hat{p} = p^*$: To increase demand or or reduce supply at p^* is not feasible without triggering a price increase for outside investors. The distinguished player i = 0 cannot gain from deviating by property (*) because $p^* < p_0$. The 2 or more shareholders who offer to buy a block of stocks cannot gain from cancelling the fill or kill order since their orders are not executed in equilibrium. Reducing demand or not trading at all is not beneficial neither. To see this note that not buying yields 0 instead of $\bar{v} - p_b = \bar{v} - p_0 + \delta - \gamma$ per share. But our specification $\delta < \frac{1}{2} \left(\frac{c}{\alpha_0} - \gamma\right)$ implies $\bar{v} - p_0 + \delta - \gamma > 0$. Increasing supply is not beneficial because it would imply selling undervalued shares.
- 2. $\hat{p} < p^*$: This can only be achieved by increasing supply. Since all shares not sold are worth $1 > p^*$ in equilibrium, this is not a profitable deviation and the distinguished player does not benefit by construction of p^* .
- 3. $\hat{p} > p^*$: An equilibrium seller *i* could deviate by using a beller strategy, buying and selling at a price $\hat{p} > p^*$. Although buying and selling some amount at the same time cancels out seller *i* gains $s_i(p^*) \cdot (\hat{p} - p^*)$ for his original equilibrium sell quantity by raising the sell price from p^* to \hat{p} . The downside is that this deviation of an equilibrium seller only works if he buys the additional supply s_j . This deviation triggers the distinguished player to sell and to reduce effort. Thereby player *i* looses $s_j(\hat{p} - \underline{v})$ since \hat{p} is what he has to pay for any such share and \underline{v} is what it is worth later on. Comparing gains and losses yields

gains =
$$s_i(p^*) \cdot (\hat{p} - p^*) \le \max_i s_i(p^*) \cdot (\hat{p} - p^*) \le s_j(\hat{p} - \underline{v}) = \text{losses}$$

since $s_j \geq \max_i s_i$ by specification of a^* and $\frac{\hat{p}-p^*}{\hat{p}-v} < 1$ since $p^* > \underline{v}$. Alternatively an equilibrium seller *i* could increase the stock price just by δ by reducing supply to $s'_i < s_i$ and not buying stocks. By this deviation seller *i* gains less than δs_i . It is not profitable, however, since sellers loose Δv on each share of their remaining stock $\alpha_i - s'_i > \alpha_i - s_i$. In our specification of a^* for every equilibrium seller holds $s_i \leq \frac{\Delta v}{\Delta v + \delta} \alpha_i \Leftrightarrow \delta s_i < \Delta v (\alpha_i - s_i)$. Thus, comparing gains and losses yields

gains
$$< \delta s_i < \Delta v (\alpha_i - s_i) < \Delta v (\alpha_i - s'_i) =$$
losses.

Furthermore, a price $\hat{p} > p^*$ could be induced by other investors who increase demand. This is not beneficial since this triggers the fill or kill order to be executed and decrease effort of the distinguished player. This is also true for the k players who submit fill or kill orders. The distinguished player's threat to sell his shares will still be viable because more than one player submits a fill or kill order.

This proves existence of an excess returns equilibrium for the Amsterdam market structure.

- (ii) Kyle market mechanism μ_K : To show existence for the Kyle market structure specify $\delta < \frac{1}{2}\left(\frac{c}{\alpha_0} \gamma\right)$ and $\alpha_0 \ge c/\Delta v$. Since the strategy of this proof is similar to the existence proof of excess returns equilibria for the Amsterdam market rules we provide a less detailed proof. Consider strategy profile a^* specified by the following conditions.
 - (1) $a_0^* = \sigma_0$ with $\sigma_0(p_0) = (\alpha_0, p_0)$: The distinguished player i = 0 submits an order using p_0 defined as in the existence proof of the Amsterdam trading rule as the limit price to sell his entire α_0 stocks.
 - (2) An investor $j \neq 0$ submits a limit order to sell less than half of what he owns, i.e. quantity $q < \frac{\alpha_j}{2}$ at limit price $p_0 \delta$.
 - (3) Each investor i = 0, 1, ..., N submits a market buy order with quantity $b_i = \frac{q}{N+1}$. Together these orders add up to $\sum_{i=0}^{N} b_i = q$, hence the entire market quantity is bought by investors who use market orders.
 - (4) Finally, it holds that $\alpha_0 c/\Delta v < q$. Investor k submits an additional buy order to buy (at least) $b_k \geq \frac{q}{N+1}$ using a price limit $p_0 \delta$.

The aggregated excess demand correspondence induced by a^* is shown in figure 2.



Figure 2: Excess returns equilibrium aggregated excess demand correspondence for the Kyle market rules

Applying the Kyle market rules to strategy profile a^* yields price $p^* = p_0 - \delta$ and $\omega_0 \Delta v = \alpha_0 \Delta v \ge c$ and in turn high effort e = 1. All equilibrium buyers except j are not sellers at the same time and therefore equilibrium winners. It remains to show that there is no strictly improving deviation for some investor. As before we organize possible deviations first by the resulting price level denoted by \hat{p} and second, among all deviations implementing price \hat{p} only consider an optimal deviation for any player i in the sense that no other deviation of this player can yield more.

1. $\hat{p} = p^*$: Deviations leading to the same price $\hat{p} = p^*$ are not beneficial. All N + 1 investors submit buy orders to buy stocks at p^* . Since the Kyle trading mechanism picks the allocation that minimizes $\sum_i (x_i)^2 + (y_i)^2$, no investor can increase the executed order size beyond $\frac{q}{N+1}$ by increasing demand at p^* . To reduce supply at p^* is not possible without triggering an increase in the stock price. Finally, the distinguished player i = 0 cannot gain from selling at p^* by construction of p_0 . Again, as for the Amsterdam rule not buying yields 0 instead of $\bar{v}-p_b = \bar{v}-p_0-\delta-\gamma$ per share. But our specification $\delta < \frac{1}{2}\left(\frac{c}{\alpha_0}-\gamma\right)$ implies $\bar{v}-p_0-\delta-\gamma > 0$.

- 2. $\hat{p} < p^*$: Picking a stock price at $\hat{p} < p^*$ is not possible for buying shareholders and not profitable for the selling shareholder.
- 3. $\hat{p} > p^*$: Deviations leading to stock prices $\hat{p} > p^*$ are clearly not beneficial for shareholders buying stocks but could potentially be beneficial for the distinguished player i = 0 or the selling shareholder i = j and achievable using a beller strategy. Since $D(p) = q = \tau(p^*)$ for all $\hat{p} > p^*$, the distinguished player cannot increase ω_0 above ω_0^* and hence to drive up the stock price $\hat{p} > p^*$ only makes buying stocks more expensive. The selling shareholder jcannot gain from picking a stock price $\hat{p} > p^*$ since j can at most sell q stocks at a price $\hat{p} > p^*$. This price increase, however, triggers the distinguished player to reduce effort to e = 0 and thus comparing j's deviation payoff with his equilibrium payoff yields

$$\hat{p} \cdot q + 0 \cdot (q - \alpha_j) \le 1 \cdot q + 0 \cdot (q - \alpha_j) < p^* \cdot q + 1 \cdot (\alpha_j - q)$$

while the last inequality holds by specification of a^* , in particular by $q < \frac{\alpha_j}{2}$. The low effort is triggered even if the selling shareholder withdraws his buy order because then investor's k buy order will be served.

- PROOF (OF THEOREM 3, PAGE 20) 1. Suppose conversely that the distinguished player i = 0 would exert low effort $e^* = 0$. Then, any equilibrium seller can improve by not selling.
 - 2. For the distinguished player i = 0 to exert effort $e^* > 0$ and to be a net seller $\omega_0^{\mu}(a^*) < \alpha_0$ is not optimal, since by not selling and exerting effort $e^* > 0$ he can improve. However, we have seen in 1. that i = 0 exerts effort $e^* > 0$. This proves 2. and shows that there must be other players being equilibrium sellers. Otherwise the trade volume would be 0.
 - 3. Suppose not. Then, any non-pivotal equilibrium seller could benefit from not selling shares since they are worth more than the equilibrium price. All non-pivotal investors could benefit from increasing demand if they could buy at the equilibrium price.
 - 4. Sellers only sell in equilibrium if they are pivotal with respect to the distinguished player's i = 0 effort choice. Since the market mechanism satisfies voluntary trade, this can only happen if the distinguished player submits an order.
 - 5. An equilibrium winner satisfies $[\bar{v} p^* \gamma] x_i^* [\bar{v} p^*] y_i^* > 0$ and violates traditional noarbitrage by definition:

$$\begin{bmatrix} \bar{v} - p^* - \gamma \end{bmatrix} x_i^* - \begin{bmatrix} \bar{v} - p^* \end{bmatrix} y_i^* > 0 \Leftrightarrow$$
$$\begin{bmatrix} p^{\mu} \left(a^* \right) - v(a^*) \end{bmatrix} y_i^{\mu} \left(a^* \right) + \begin{bmatrix} v(a^*) - p^{\mu} \left(a^* \right) - \gamma \end{bmatrix} x_i^{\mu} \left(a^* \right) > 0.$$

PROOF (OF PROPOSITION 1, PAGE 21) Let $\Delta v = 0$. This means that $e^* = 0$ which is a contradiction for a^* to be an excess returns equilibrium due to Theorem 3 part 1.

PROOF (OF PROPOSITION 2, PAGE 21) 1. Consider the converse case of a high true value equilibrium with bid ask spread $\gamma > 0$ where no outside investor $k \in \{1, ..., N\}$ is pivotal in the sense that selling less or buying more than specified by equilibrium strategies triggers the distinguished player i = 0 to sell and to reduce effort subsequently. In this case any equilibrium buyer can improve by not submitting buy orders and thereby saving $\gamma > 0$.

- 2. In a low true value trade equilibrium with $p^* = \underline{v}$ with bid ask spread $\gamma > 0$ a buyer could improve by not buying since he pays $p_b^* = \underline{v} + \gamma$. In contrast to the high true value equilibrium pivotalness does not play a role since by definition players payoff cannot decrease below \underline{v} per share. Since a low true value equilibrium does not exist, every equilibrium involves $e^* > 0$. For this to occur, the distinguished player must buy shares if $\alpha_0 < c/\Delta v$. But then, the distinguished player can improve by not buying (and saving effort costs and bid ask spread), a contradiction.
- 3. For a true value equilibrium a^* holds $p^{\mu}(a^*) = v(a^*)$ and thereby

$$\left[p^{\mu} \left(a^{*} \right) - v(a^{*}) \right] y_{i}^{\mu} \left(a^{*} \right) + \left[v(a^{*}) - p^{\mu} \left(a^{*} \right) - \gamma \right] x_{i}^{\mu} \left(a^{*} \right) = 0 \cdot y_{i}^{*} - \gamma \cdot x_{i}^{*} \leq 0$$

since $\gamma \cdot x_i^* \ge 0$ by definition.

4. Construct a strategy profile with resulting net trade $x - y \neq 0$ such that no player can strictly improve by deviating. To do this let a player j with positive initial stake $\alpha_j > 0$ just submit a single limit sell order $a_j = \{\sigma_j\} = \{(s_j, \bar{v})\}$. Moreover, the constraints $s_j \leq \alpha_j$ and $s_j \leq \alpha_0 - \frac{c}{\Delta v}$ must be satisfied. Since either quantities are continuous or M is large enough, we can find quantities that satisfy the latter inequality. Some other player -j submits a single limit buy order $a_{-j} = \{\beta_{-j}\} = \{(s_j, \bar{v})\}$. And all other players submit nothing. If μ maximizes the trade volume at equilibrium price $p^* = \bar{v}$ then quantity s_j is traded between player j and player -j and therefore we observe a non-zero net transaction $x - y \neq 0$. No player can improve since for any other price players can only trade with themselves and at true value equilibrium price $p^* = \bar{v}$ they are indifferent between trading and not trading. Especially, the distinguished player i = 0 cannot sell enough to choose e = 0 subsequently. This proves existence for a trade volume maximizing market mechanism μ .

Consider now $\alpha_0 \Delta v < c$ and $\underline{v} > 0$. This implies $\omega_0 \Delta v < c$ and a low effort level e = 0. This implies that if a true value equilibrium exists then it yields low price $p^* = \underline{v}$. Existence is constructed similarly as before by picking a pair of players submitting one buy order and one sell order at this price but nothing else. Again, a trade volume maximizing market mechanism guarantees that the pair of orders is executed at this price and nobody can improve. If quantities are continuous or M is sufficiently large, we can again find traded quantities that do not allow the distinguished player to buy shares and exert effort subsequently.

PROOF (OF THEOREM 4, PAGE 24) Structure of the proof. We proceed by defining an economy with ownership structure and budget constraints. For this economy, we propose a candidate equilibrium strategy profile a^* for the distinguished player and rational investors. The task is to specify everything appropriately such that one finds a strategy profile a^* that establishes an excess returns equilibrium. The strategies, ownership structure, and budget constraints we propose are supposed to satisfy conditions (2)-(4). The proof contains three auxiliary results lemmata 3 to 5. While lemma 3 states some preparatory results lemma 4 shows that there exist parameters fulfilling conditions (2)-(4). Finally, lemma 5 shows that for candidate equilibrium strategy profile a^* there are no strictly improving deviations and thereby that indeed it forms a Nash equilibrium given the parameters satisfy conditions (2)-(4). The proofs in part (i) for μ_N and parts (ii) and (iii) for μ_S and μ_T only differ in the specification of the distinguished player's equilibrium strategy and in the budget constraints.

(i) 1. The economy. Small rational outside investors²⁹ are specified with initial stake $\alpha_i \ \forall i \in$

²⁹When we talk about the stakes of small continuum investors we mean infinitesimal stakes. For example, a measurable subset J of small investors jointly owns $\int_J \alpha_i di = \int_J 1 d\alpha$ stakes. For example, if J has Lebesgue measure $\lambda(J)$ and every investor owns the same α_r infinitesimal stakes then they jointly own $\lambda(J) \cdot \alpha_r$ stakes. The same holds for infinitesimal budgets.

(0,1) with aggregated stake $\int_{(0,1)} \alpha_i di = \alpha_r$ and budget constraint B_i with aggregated constraint $\int_{(0,1)} B_i di = B > \overline{v}$. This specification implies that the aggregated budget constraint is never binding since together small investors can afford to buy strictly more than the entire firm at the highest value. However for any single investor there is a finite upper bound B_i up to which buy orders can be submitted.³⁰

2. Equilibrium strategies.

DP Distinguished player

 $a_0^* = \sigma_0$ with $\sigma_0(p_0) = (\alpha_0, p_0)$ where

$$p_0 := \min_{p \in P} \left\{ p | p > \bar{v} - \frac{c}{\alpha_0} \right\}.$$

$$\tag{2}$$

The distinguished player i = 0 submits a single order using p_0 as the limit price to sell his shares. This first condition specifies the price limit of the distinguished player as the lowest price at which the distinguished player has a strict incentive to sell instead of exerting effort.

ROI Rational outside investors

 $a_{i}^{*} = \{\beta_{i}^{*}, \sigma_{i}^{*}\} = \{(\rho_{i}, p_{0} - \delta), (\alpha_{i}, p_{0})\} \text{ for } i \neq 0 \text{ with } \rho_{i} := \frac{B_{i}}{p_{0} - \delta}.$

Here, ρ_i is the maximal buy order at price $p_0 - \delta$ any rational buyer can afford according to his budget constraint. Outside investors submit maximal buy orders for ρ_i shares using $p_0 - \delta$ as a price limit and sell orders for all their shares using p_0 as the limit price.

3. Allocation and prices

The market mechanism together with the action profile specifies the expected equilibrium price, the value of the firm – implicitly determined by the expost stakes owned by the distinguished player – and which outside investors' orders are executed determined by stochastic rationing.

Expected price E(p).

In this strategy profile rational investors and the distinguished player never trade with each other. Therefore, market mechanism $\mu = \mu_N$ picks $p^{\mu} = p_0 - \delta$ if $\tilde{Z}_1 < 0$ and $p^{\mu} = p_0$ if $\tilde{Z}_1 > 0$. This yields an expected price

$$E(p) = \Pr(\tilde{Z}_1 < 0) \cdot (p_0 - \delta) + \Pr(\tilde{Z}_1 \ge 0) \cdot p_0.$$

Expected company value E(v).

The expected value of the company depends only on the distinguished player's final stake ω_0 . Since the market mechanism guarantees fully executed orders, the distinguished player either sells all his shares or none at all. Consequently, the distinguished player exerts effort if $\omega_0 > 0$. This implies an expected value

$$E(v) = \Pr(\omega_0 = \alpha_0) \cdot \bar{\nu} + \Pr(\omega_0 \neq \alpha_0) \cdot \underline{v}.$$

³⁰The following interpretation of the budget constraint matters for the proof. Though B_i can be arbitrarily large, we suppose that the market mechanism does not execute buy orders $\beta_i = (b, p)$ with $b \cdot p > B_i$ of small investors since they could not not afford them unless they submit sell orders that are executed. Since traders don't know in advance if and for which price their sell order is executed – which could be random – buy orders are not allowed to be based on them. This assumption excludes equilibrium deviations where traders behave as bellers – i.e. buyers and sellers at the same time in order to increase their budget.

It is important to note that $Pr(\omega_0 \neq \alpha_0) > 0$ as long as $\alpha_0 < b$, i.e. α_0 is small enough such that the distinguished player can sell all his shares to noise traders. To guarantee an *excess* returns equilibrium we state the second condition

$$E(v^*) - E(p^*) =$$

 $R(p_0) > 0$ (3)

where

$$R(p) := \Pr(\tilde{Z}_1 < 0) \cdot (\bar{\nu} - p + \delta) + \Pr(\tilde{Z}_1 \ge 0) \cdot \{\Pr(\omega_0 = \alpha_0 | \tilde{Z}_1 > 0) \cdot (\bar{\nu} - p) + \Pr(\omega_0 \neq \alpha_0 | \tilde{Z}_1 > 0) \cdot (\underline{\nu} - p)\}.$$

Note that R(p) is a measure for excess returns.

Rationing.

To determine the rationing parameters consider the 3 relevant events:

(a) Noise traders sell in aggregate with probability $Pr(\tilde{Z}_1 < 0)$: In this case the market mechanism realizes price $p_0 - \delta$ and aggregated demand is $\rho = \int_{(0,1)} \rho_i$ while expected aggregated supply is $E(\|\tilde{Z}_1\| | \tilde{Z}_1 < 0)$. This implies that every rational buyer will be served with probability

$$\lambda_1 := \frac{E(\|\ddot{Z}_1\| | \ddot{Z}_1 < 0)}{\rho}.$$

Since $\rho > 1$ and $0 < E(\|\tilde{Z}_1\| | \tilde{Z}_1 < 0) < 1$ there will always be rationing $0 < \lambda_1 < 1$ in this event.

(b) Noise traders buy in aggregate and the distinguished player does not sell his shares which occurs with probability $Pr(\tilde{Z}_1 \ge 0) \cdot Pr(\omega_0 = \alpha_0 | \tilde{Z}_1 \ge 0)$: In this case price p_0 is realized and expected aggregated demand is $E(\|\tilde{Z}_1\| | \tilde{Z}_1 \ge 0)$ while aggregated supply is α_r . Every small outside investor is served with probability

$$\lambda_2 := \frac{E(\|\tilde{Z}_1\||\tilde{Z}_1 \ge 0 \text{ and } \omega_0 = \alpha_0)}{\alpha_r}$$

(c) Noise traders buy in aggregate and the distinguished player sells his shares which occurs with probability $\Pr(\tilde{Z}_1 \ge 0) \cdot \Pr(\omega_0 \ne \alpha_0 | \tilde{Z}_1 \ge 0)$: In this case price p_0 is realized and expected aggregated demand is $E(\|\tilde{Z}_1\| | \tilde{Z}_1 \ge 0)$ which has to be reduced by α_0 since the distinguished player is assumed to sell his shares. Again, aggregated supply is α_r from rational investors. Every small outside investor is served with probability

$$\lambda_3 := \frac{E(\|Z_1\| | Z_1 \ge 0 \text{ and } \omega_0 \neq \alpha_0) - \alpha_0}{\alpha_r}$$

The third condition reflects the critical trade-off for outside investors. It makes sure that they do not want to deviate from their equilibrium strategy and is given as

$$\Pr\left(\tilde{Z}_1 < 0\right) \cdot \lambda_1 \cdot (\bar{\nu} - p_0 + \delta) \ge R(p_0). \tag{4}$$

Investors compare being rationed in their buying order but only buying for the beneficial lower price $p_0 - \delta$ against always being served but including situations where they can trade against the distinguished player. The crucial difference here to the remainder of the article is that small traders can neither influence the price – i.e. they are *price takers* – nor the final allocation of other traders or the firm value and hence they are *not pivotal*.

The following claims are preparatory results that help to show existence in lemma 4.

Lemma 3 Part A: Let $\Pi(p) := R(p) - \Pr\left(\tilde{Z}_1 < 0\right) \cdot \lambda_1 \cdot (\bar{\nu} - p + \delta)$. For small enough δ , $R(\underline{v}) > 0 > R(\bar{v})$ and $\Pi(\underline{v}) > 0 > \Pi(\bar{v})$ if $\Pr\left(\omega_0 \neq \alpha_0 | \tilde{Z}_1 > 0\right) > 0$. Part B: Conditions (2), (3), and (4) imply the following:

- 1. $p_0 > \underline{v}$.
- 2. For small enough δ the IC constraint $c < \alpha_0 \Delta v$ is fulfilled
- 3. $p_0 > E(v|\tilde{Z}_1 > 0)$: Shares are overvalued if $p^* = p_0$.

PROOF (OF LEMMA 3) Part A: $R(\underline{v}) > 0$ follow from plugging $p = \underline{v}$ into the definition of R(p). $R(\overline{v}) < 0$: This is true for small enough δ if $\Pr\left(\omega_0 \neq \alpha_0 | \tilde{Z}_1 > 0\right) > 0$ because

$$R(\bar{v}) = \Pr(\tilde{Z}_1 < 0) \cdot \delta + \Pr\left(\omega_0 \neq \alpha_0 | \tilde{Z}_1 > 0\right) \cdot (\underline{v} - \bar{v}) < 0$$

holds for small enough δ . To prove the second set of inequalities, replace $R(\cdot)$ in above arguments by $\Pi(\cdot)$. This proves the claim.

Part B:

Claim 1: Suppose not. Then, it follows from the definition of $R(p_0)$ that $R(p_0) > Pr(\tilde{Z}_1 < 0) \cdot (\bar{\nu} - p_0 + \delta) > Pr(\tilde{Z}_1 < 0) \cdot \lambda_1 \cdot (\bar{\nu} - p_0 + \delta)$ where the latter inequality is a contradiction of (4). Claim 2: From equation (2) it follows that $\bar{v} - \frac{c}{\alpha_0} > p_0 - \delta$. Hence, if $p_0 - \delta > \underline{v} \Leftrightarrow p_0 - \underline{v} > \delta$ (which is possible due to lemma 3 claim 1) it is true that $\bar{v} - \frac{c}{\alpha_0} > \underline{v} \Leftrightarrow \alpha_0 \ge c/\Delta v$. This proves the claim.

Claim 3. Suppose not \Rightarrow

$$E\left(v|\tilde{Z}_1>0\right) - p_0 =$$

$$\Pr(\omega_0 = \alpha_0|\tilde{Z}_1>0) \cdot (\bar{\nu} - p_0) + \Pr\left(\omega_0 \neq \alpha_0|\tilde{Z}_1>0\right) \cdot (\underline{v} - p_0) \ge 0 \Leftrightarrow$$

$$R(p_0) - \Pr\left(\tilde{Z}_1<0\right) \cdot (\bar{\nu} - p_0 + \delta) \ge 0$$

which by $\lambda_1 < 1$ implies $R(p_0) > \Pr(\tilde{Z}_1 < 0) \cdot \lambda_1 \cdot (\bar{\nu} - p_0 + \delta)$, a contradiction to inequality (3).

With these observations we can now show existence:

Lemma 4 For small enough δ , there always exist p_0, α_0 , and c such that constraints (2), (3), and (4) hold for any non-degenerate noise.

PROOF (OF LEMMA 4) In a first step we choose α_0 small enough to ensure $\Pr(\omega_0 \neq \alpha_0 | \tilde{Z}_1 > 0) > 0$. We can now find a p_0 to guarantee that $R(p_0)$ is positive and sufficiently small since $R(\underline{v}) > 0 > R(\overline{v})$ (lemma 3) and R is continuous in p. Note that $p_0 < \overline{v}$ (since otherwise $R(p_0) < 0$). Hence, it is true that $\Pr\left(\tilde{Z}_1 < 0\right) \cdot \lambda_1 \cdot (\overline{v} - p_0 + \delta) > 0$. Small enough but positive excess returns $R(p_0)$ together with the observation that $\Pr\left(\tilde{Z}_1 < 0\right) \cdot \lambda_1 \cdot (\overline{v} - p_0 + \delta) > 0$ guarantee equations (3) and (4). Constraint (2) can be guaranteed by choosing c appropriately.

Lemma 5 Strategy $a^* = \{a_0^*, \{a_i^*\}_{i \in (0,1)}\}$ with $a_0^* = \sigma_0(p_0) = (\alpha_0, p_0)$ and $a_i^* = \{\beta_i^*, \sigma_i^*\} = \{(\rho_i, p_0 - \delta), (\alpha_i, p_0)\}$ for $i \neq 0$ is an equilibrium strategy if constraints (2), (3), and (4) are met and $\mu = \mu_N$.

PROOF (OF LEMMA 5) We first derive equilibrium utility and then check possible deviations.

1. Equilibrium utility.

Equilibrium utility of a risk neutral rational outside investor $i \in (0, 1)$ is given by $u_i(a^*) = E[\omega_i \tilde{v} - p\tilde{x}_i + p\tilde{y}_i]$ where now the company value \tilde{v} and the executed trade vector $(\tilde{x}_i, \tilde{y}_i)$ are random variables depending on the realization of noise \tilde{Z}_1 . For the given strategy profile

$$u_{i}(a^{*}) = E\left[\omega_{i}\tilde{v} - p\tilde{x}_{i} + p\tilde{y}_{i}\right]$$

$$= \Pr(\tilde{Z}_{1} < 0) \cdot \left[\left(\alpha_{i} + \lambda_{1}\rho_{i}\right)\bar{v} - \lambda_{1}\rho_{i}(p_{0} - \delta)\right]$$

$$+ \Pr(\omega_{0} = \alpha_{0}|\tilde{Z}_{1} > 0) \cdot \alpha_{i} \cdot \left[\left(1 - \lambda_{2}\right) \cdot \bar{\nu} + \lambda_{2} \cdot p_{0}\right]$$

$$+ \Pr\left(\omega_{0} \neq \alpha_{0}|\tilde{Z}_{1} > 0\right) \cdot \alpha_{i} \cdot \left[\left(1 - \lambda_{3}\right) \cdot \underline{\nu} + \lambda_{3} \cdot p_{0}\right].$$

In the first expression $\lambda_1 \rho_i (\bar{v} - p_0 + \delta)$ are the expected benefits from buying shares at price $p_0 - \delta$ if $\tilde{Z}_1 < 0$. The second and third expressions are utility from selling shares which can happen if $\tilde{Z}_1 \geq 0$. The second term is utility from selling if the distinguished player exerts high effort. Finally, if the distinguished player sells his shares he will not exert effort. Again, there is rationing and only a fraction λ_3 of all sell orders can be served in that case.

2. Deviations.

We discuss deviations at any equilibrium price that can occur with positive probability.

- (a) Price $p \ge p_0$.
 - i. Outside investor $i \in (0, 1)$: No outside investor has an impact on equilibrium price and we can restrict our analysis of deviations to the two realizations of equilibrium prices. Outside investors could increase their demand at $p \ge p_0$ by submitting a buy orders $\beta'_i = (b, p_0 + \epsilon)$ with b > 0 and $\epsilon \ge 0$.

As a consequence, the buy order would not be rationed due to price priority which is beneficial for the low price and adverse if the distinguished player sells his shares. It is not a beneficial deviation if

$$\begin{aligned} \Pr(\tilde{Z}_1 < 0) \cdot \lambda_1 \cdot (\bar{\nu} - p_0 + \delta) &\geq & \Pr(\tilde{Z}_1 < 0) \cdot (\bar{\nu} - p_0 + \delta) \\ &+ \Pr(\omega_0 = \alpha_0 | \tilde{Z}_1 > 0) \cdot (\bar{\nu} - p_0) \\ &+ \Pr\left(\omega_0 \neq \alpha_0 | \tilde{Z}_1 > 0\right) \cdot (\underline{\nu} - p_0) \end{aligned}$$

which holds by equation (4).

Decreasing sell orders is not beneficial since shares are overvalued if $p \ge p_0$ by virtue of lemma 3 part 3.

- ii. Distinguished Player: Offering to sell at higher prices is not beneficial since these orders will not be executed. This is true because we assume that $b \leq \alpha_{\rho}$ and hence $\tilde{Z}_1 > \alpha_{\rho}$ is not possible and orders from rational outside investors can always match the noise. Offering to sell only a fraction is not beneficial by construction of p_0 .
- (b) Price $p < p_0$.
 - i. Outside investor $i \in (0, 1)$: Clearly, increasing supply or decreasing demand at $p < p_0$ is not a beneficial deviation since shares are undervalued for $p < p_0$.
 - ii. Distinguished Player: Selling at $p < p_0$ is not beneficial by construction of p_0 .

This completes the proof of part (i).

(ii) $\mu = \mu_S$: Specify similarly as in (i). However, the equilibrium strategy of the distinguished player is a fill-or-kill order: $a_0^* = \mathring{\sigma}_0$ with $\mathring{\sigma}_0(p_0) = (\alpha_0, p_0)$. The distinguished player i = 0 submits a single fill-or-kill order using the same p_0 as the limit price to sell all his shares. Furthermore, budget constraints are symmetric: $b_i = b_j$ for all $i \neq j$ and $i \in (0, 1)$ and $j \in (0, 1)$.

- (iii) $\mu = \mu_T$: Specify noise $Pr(\tilde{Z}_1 \in (0, t) = 0)$ for some t > 0 and $\alpha_0 \le t$. Furthermore, budget constraints are as in (ii). Finally, the distinguished player now submits the same limit order as in (i). With these re-specifications the proof is the same as in (i).
- (iv) The proof makes use of the following observations. Condition (2) is not affected by a change of the distribution of noise. However, condition (4) is harder to meet if noise decreases. Hence, we construct a strategy profile a^* together with p_0 and c such that condition (4) is satisfied under high noise but not under low noise.

Let p_h be implicitly defined by $\Pi_l(p_h) = 0$. Clearly, p_h exists since $\Pi(\underline{v}) > 0 > \Pi(\overline{v})$ and $\Pi(p)$ is continuous in p. Now, pick an $\{\alpha_0, c, a^*\}$ for the economy specified in part (i) of this proof with p_0 sufficiently close to p_h which satisfies conditions (2), (3), and (4). By lemma 5 there exist such parameters that a^* is an equilibrium under noise h. However, by construction these parameters cease to be an equilibrium under noise l since rational investors have an incentive to deviate from their proposed equilibrium strategies because condition (4) no longer holds if p_0 is sufficiently close to p_h . To see this, note that λ_1 strictly decreases if noise is reduced from high noise h to low noise l. Hence, the LHS of condition (4) strictly decreases. Furthermore, the RHS of condition (4) cannot decrease and may even increase.

This completes the proof of part (iv).

- (v) Without noise, our call auction market mechanisms obey the voluntary trade property. Hence, we can again apply our characterization result. However, since agents cannot be pivotal in the continuum case, an excess returns equilibrium cannot exist.
- (vi) The proof is by contradiction. Suppose to the contrary that there exists an excess returns equilibrium. This implies that shares are traded at a price $p < v = \bar{v} = \underline{v}$. Note first that not every investor can be served with probability one since $\alpha_1 < 1$ but aggregate budget constraint allows rational investors to buy more than the entire firm at prices p < v. Therefore, the strictly positive measure of traders who are not served with probability one have an incentive to increase their price limit. This would increase their equilibrium utility since they will now always be served at p by virtue of the price priority property of the market microstructures studied in this section.

This completes the proof of theorem 4.

References

Aitken, M., A. Frino, M. McCorry and P. Swan (1998): "Short Sales Are Almost Instantaneously Bad News: Evidence from the Australian Stock Exchange." *The Journal* of *Finance*, 53(6), 2205-2223.

Admati, A., P. Pfleiderer and J. Zechner (1994), "Large Shareholder Activism, Risk Sharing, and Financial Market Equilibrium" *The Journal of Political Economy* 102(6), 1097-1130

Bagnoli, M. and B. Lipman (1988), "Successful Takeovers without Exclusion". *Review* of Financial Studies, 89-107

Biais, B, P. Hillion, and C. Spatt (1995): "An Empirical Analysis of the Limit Order Book and the Order Flow in the Paris Bourse." *The Journal of Finance*, 50(5), 1655-1689.

Bisin, A. and D. Guaitoli (2004): "Moral hazard and nonexclusive contracts." *RAND Journal of Economics.* 35(2), 306-328.

Bizer D. S. and P. M. DeMarzo (1992): "Sequential Banking." *The Journal of Political Economy*. 100(1), 41-61.

Blonski, M. (2005) "*Games in Aggregated Form*", working paper, Johann Wolfgang Goethe Universität, Frankfurt am Main

Bolton, P. and E. von Thadden (1998): "Blocks, Liquidity, and Corporate Control", *The Journal of Finance.* 53(1), 1-25.

Brunnermeier, M. (2001): "Asset pricing under asymmetric information: bubbles, crashes, technical analysis, and herding." Oxford: Oxford Univ. Press.

Chan, Louis K. C. and Josef Lakonishok (1995): "The Behavior of Stock Prices around Institutional Trades." *Journal of Finance*, 50(4), 1147-74.

DeMarzo, P. M. and B. Urosevic (2006): "Ownership Dynamics and Asset Pricing with a Large Shareholder." *Journal of Political Economy*, 114(4), 774-815

Fahlenbrach, Rüdiger (2007): "Founder-CEOs, Investment Decisions, and Stock Market Performance", *The Journal of Financial and Quantitative Analysis*, forthcoming.

Fahlenbrach, Rüdiger and Rene M. Stulz (2007): *Managerial Ownership Dynamics* and *Firm Value*, National Bureau of Economic Research, Inc, NBER Working Papers: 13202.

Fama, E. F., (1970) "Efficient Capital Markets: A Review of Theory and Empirical Work," *Journal of Finance*, 25(2), 383-417.

Gompers, Paul A. and Andrew Metrick (2001): "Institutional Investors and Equity Prices" *Quarterly Journal of Economics*, 116(1), 229-59.

Grossman, S. J. and O. Hart (1980): "Takeover Bids, the free-rider problem, and the theory of the corporation." *The Bell Journal of Economics*, 11, 42-64.

Grossman, S. J. and O. Hart (1983): "An Analysis of the Principal-Agent Problem.", *Econometrica*. 51(1), 7-46.

Grossman, S. J. and J. E. Stiglitz (1980): "On the Impossibility of Informationally Efficient Markets", *The American Economic Review*, 70(3), 393-408.

Holderness, Clifford G. (2007): "The Myth of Diffuse Ownership in the United States", *Review of Financial Studies.*, forthcoming.

Holmström, B (1979): "Moral Hazard and Observability", *The Bell Journal of Economics*. 10(1), 74-9.

Holmström, B. and B. Nalebuff (1992): "To the Raider Goes the Surplus? A Reexamination of the Free-Rider Problem." *The Journal of Economics and Management Strategy*, 1(1), 37-62.

Huang, R. and H. Stoll (2001): "Tick Size, Bid-Ask Spreads and Market Structure", Journal of Financial and Quantitative Analysis, 36(4), 503-522.

Kahn, C. and D. Mookherjee (1998): "Competition and incentives with nonexclusive contracts." *RAND Journal of Economics.* 29(3), 443-465.

Kahn, C. and A. Winton (1998): "Ownership Structure, Speculation, and Shareholder Intervention." *The Journal of Finance*. 53(1), 99-129.

Kalai, E. (2004) "Large robust games", Econometrica 72, 1631-1665.

Kyle, A. S. (1985): "Continuous Auctions and Insider Trading." *Econometrica*, 53 (6), 1315-35.

Kyle, A. S. (1989): "Informed Speculation with Imperfect Competition.", *The Review of Economic Studies*. 56 (3), 317-55.

O'Hara, Maureen (1995): "Market microstructure theory." Blackwell, Malden, Mass.
O'Hara, Maureen (2003): "Presidential Address: Liquidity and Price Discovery." Journal of Finance, 58 (4), 1335-54.

Lehmann, Bruce N. and David M. Modest (1994): "Trading and Liquidity on the Tokyo Stock Exchange: A Bird's Eye View". *Journal of Finance*, 49(3), 951-84.

von Lilienfeld-Toal, U. (2005): "Why Managers Hold Shares of Their Firm: Theory" working paper, Johann Wolfgang Goethe Universität, Frankfurt am Main

von Lilienfeld-Toal, U. and S. Rünzi (2007): "Why Managers Hold Shares of Their Firm: An Empirical Analysis." working paper, Johann Wolfgang Goethe Universität, Frankfurt am Main.

Magill, M. and M. Quinzii (2002): "Capital Market Equilibrium with Moral Hazard." *Journal of Mathematical Economics*, 38(1-2), 149-90.

Maug, E. (1998): "Large Shareholders as Monitors: Is There a Trade-off between Liquidity and Control?" *The Journal of Finance*. 53(1), 65-98.

Milgrom, P. and N. Stokey (1982): "Information, Trade and Common Knowledge." *The Journal of Economic Theory.* 26(1), 17-27.

Reny, P. and M. Perry (2006): Toward a Strategic Foundation for Rational Expectations Equilibrium, *Econometrica*, 74(5), 1231–1269.

Rochet, J.-C. and J.-L. Vila (1994): "Insider Trading without Normality." *The Review* of *Economic Studies*. 61(1), 131-52.

Ross, S., (1976): "The arbitrage theory of capital asset pricing", *Journal of Economic Theory.* 13 (3)

Segal, I. and M.D.Whinston (2004), "Robust Predictions for Bilateral Contracting with Externalities", *Econometrica*. 71(3), 757-91

Shleifer, A. and R. Vishny (1986): "Large Shareholders and Corporate Control." *The Journal of Political Economy*, 94(3), 461-488.

Tirole, J. (1982): "On the Possibility of Speculation under Rational Expectations." *Econometrica*. 50(5), 1163-81.

Venkataraman, K. (2001): Automated versus Floor Trading: An analysis of execution costs on the Paris and New York Exchanges. *Journal of Finance*, 56(4), 1445-1885.

Xetra (2003): XETRA Release 6.0, Market Model Stock Trading.

Bisher erschienene Diskussionspapiere

- Nr. 78: Blonski, Matthias; Lilienfeld-Toal, Ulf von: Excess Returns and the Distinguished Player Paradox, Oktober 2008
- Nr. 77: Lechner, Susanne; Ohr, Renate: The Right of Withdrawal in the Treaty of Lisbon: A game theoretic reflection on different decision processes in the EU, Oktober 2008
- Nr. 76: Kühl, Michael: Strong comovements of exchange rates: Theoretical and empirical cases when currencies become the same asset, Juli 2008
- Nr. 75: Höhenberger, Nicole; Schmiedeberg, Claudia: Structural Convergence of European Countries, Juli 2008
- Nr. 74: Nowak-Lehmann D., Felicitas; Vollmer, Sebastian; Martinez-Zarzoso, Inmaculada: Does Comparative Advantage Make Countries Competitive? A Comparison of China and Mexico, Juli 2008
- Nr. 73: Fendel, Ralf; Lis, Eliza M.; Rülke, Jan-Christoph: Does the Financial Market Believe in the Phillips Curve? Evidence from the G7 countries, Mai 2008
- Nr. 72: Hafner, Kurt A.: Agglomeration Economies and Clustering Evidence from German Firms, Mai 2008
- Nr. 71: Pegels, Anna: Die Rolle des Humankapitals bei der Technologieübertragung in Entwicklungsländer, April 2008
- Nr. 70: Grimm, Michael; Klasen, Stephan: Geography vs. Institutions at the Village Level, Februar 2008
- Nr. 69: Van der Berg, Servaas: How effective are poor schools? Poverty and educational outcomes in South Africa, Januar 2008
- Nr. 68: Kühl, Michael: Cointegration in the Foreign Exchange Market and Market Efficiency since the Introduction of the Euro: Evidence based on bivariate Cointegration Analyses, Oktober 2007
- Nr. 67: Hess, Sebastian; Cramon-Taubadel, Stephan von: Assessing General and Partial Equilibrium Simulations of Doha Round Outcomes using Meta-Analysis, August 2007
- Nr. 66: Eckel, Carsten: International Trade and Retailing: Diversity versus Accessibility and the Creation of "Retail Deserts", August 2007
- Nr. 65: Stoschek, Barbara: The Political Economy of Environmental Regulations and Industry Compensation, Juni 2007
- Nr. 64: Martinez-Zarzoso, Inmaculada; Nowak-Lehmann D., Felicitas; Vollmer, Sebastian: The Log of Gravity Revisited, Juni 2007
- Nr. 63: Gundel, Sebastian: Declining Export Prices due to Increased Competition from NIC Evidence from Germany and the CEEC, April 2007
- Nr. 62: Wilckens, Sebastian: Should WTO Dispute Settlement Be Subsidized?, April 2007
- Nr. 61: Schöller, Deborah: Service Offshoring: A Challenge for Employment? Evidence from Germany, April 2007
- Nr. 60: Janeba, Eckhard: Exports, Unemployment and the Welfare State, März 2007
- Nr. 59: Lambsdoff, Johann Graf; Nell, Mathias: Fighting Corruption with Asymmetric Penalties and Leniency, Februar 2007
- Nr. 58: Köller, Mareike: Unterschiedliche Direktinvestitionen in Irland Eine theoriegestützte Analyse, August 2006
- Nr. 57: Entorf, Horst; Lauk, Martina: Peer Effects, Social Multipliers and Migrants at School: An International Comparison, März 2007 (revidierte Fassung von Juli 2006)

- Nr. 56: Görlich, Dennis; Trebesch, Christoph: Mass Migration and Seasonality Evidence on Moldova's Labour Exodus, Mai 2006
- Nr. 55: Brandmeier, Michael: Reasons for Real Appreciation in Central Europe, Mai 2006
- Nr. 54: Martínez-Zarzoso, Inmaculada; Nowak-Lehmann D., Felicitas: Is Distance a Good Proxy for Transport Costs? The Case of Competing Transport Modes, Mai 2006
- Nr. 53: Ahrens, Joachim; Ohr, Renate; Zeddies, Götz: Enhanced Cooperation in an Enlarged EU, April 2006
- Nr. 52: Stöwhase, Sven: Discrete Investment and Tax Competition when Firms shift Profits, April 2006
- Nr. 51: Pelzer, Gesa: Darstellung der Beschäftigungseffekte von Exporten anhand einer Input-Output-Analyse, April 2006
- Nr. 50: Elschner, Christina; Schwager, Robert: A Simulation Method to Measure the Tax Burden on Highly Skilled Manpower, März 2006
- Nr. 49: Gaertner, Wulf; Xu, Yongsheng: A New Measure of the Standard of Living Based on Functionings, Oktober 2005
- Nr. 48: Rincke, Johannes; Schwager, Robert: Skills, Social Mobility, and the Support for the Welfare State, September 2005
- Nr. 47: Bose, Niloy; Neumann, Rebecca: Explaining the Trend and the Diversity in the Evolution of the Stock Market, Juli 2005
- Nr. 46: Kleinert, Jörn; Toubal, Farid: Gravity for FDI, Juni 2005
- Nr. 45: Eckel, Carsten: International Trade, Flexible Manufacturing and Outsourcing, Mai 2005
- Nr. 44: Hafner, Kurt A.: International Patent Pattern and Technology Diffusion, Mai 2005
- Nr. 43: Nowak-Lehmann D., Felicitas; Herzer, Dierk; Martínez-Zarzoso, Inmaculada; Vollmer, Sebastian: Turkey and the Ankara Treaty of 1963: What can Trade Integration Do for Turkish Exports, Mai 2005
- Nr. 42: Südekum, Jens: Does the Home Market Effect Arise in a Three-Country Model?, April 2005
- Nr. 41: Carlberg, Michael: International Monetary Policy Coordination, April 2005
- Nr. 40: Herzog, Bodo: Why do bigger countries have more problems with the Stability and Growth Pact?, April 2005
- Nr. 39: Marouani, Mohamed A.: The Impact of the Mulitfiber Agreement Phaseout on Unemployment in Tunisia: a Prospective Dynamic Analysis, Januar 2005
- Nr. 38: Bauer, Philipp; Riphahn, Regina T.: Heterogeneity in the Intergenerational Transmission of Educational Attainment: Evidence from Switzerland on Natives and Second Generation Immigrants, Januar 2005
- Nr. 37: Büttner, Thiess: The Incentive Effect of Fiscal Equalization Transfers on Tax Policy, Januar 2005
- Nr. 36: Feuerstein, Switgard; Grimm, Oliver: On the Credibility of Currency Boards, Oktober 2004
- Nr. 35: Michaelis, Jochen; Minich, Heike: Inflationsdifferenzen im Euroraum eine Bestandsaufnahme, Oktober 2004
- Nr. 34: Neary, J. Peter: Cross-Border Mergers as Instruments of Comparative Advantage, Juli 2004
- Nr. 33: Bjorvatn, Kjetil; Cappelen, Alexander W.: Globalisation, inequality and redistribution, Juli 2004
- Nr. 32: Stremmel, Dennis: Geistige Eigentumsrechte im Welthandel: Stellt das TRIPs-Abkommen ein Protektionsinstrument der Industrieländer dar?, Juli 2004
- Nr. 31: Hafner, Kurt: Industrial Agglomeration and Economic Development, Juni 2004

- Nr. 30: Martinez-Zarzoso, Inmaculada; Nowak-Lehmann D., Felicitas: MERCOSUR-European Union Trade: How Important is EU Trade Liberalisation for MERCOSUR's Exports?, Juni 2004
- Nr. 29: Birk, Angela; Michaelis, Jochen: Employment- and Growth Effects of Tax Reforms, Juni 2004
- Nr. 28: Broll, Udo; Hansen, Sabine: Labour Demand and Exchange Rate Volatility, Juni 2004
- Nr. 27: Bofinger, Peter; Mayer, Eric: Monetary and Fiscal Policy Interaction in the Euro Area with different assumptions on the Phillips curve, Juni 2004
- Nr. 26: Torlak, Elvisa: Foreign Direct Investment, Technology Transfer and Productivity Growth in Transition Countries, Juni 2004
- Nr. 25: Lorz, Oliver; Willmann, Gerald: On the Endogenous Allocation of Decision Powers in Federal Structures, Juni 2004
- Nr. 24: Felbermayr, Gabriel J.: Specialization on a Technologically Stagnant Sector Need Not Be Bad for Growth, Juni 2004
- Nr. 23: Carlberg, Michael: Monetary and Fiscal Policy Interactions in the Euro Area, Juni 2004
- Nr. 22: Stähler, Frank: Market Entry and Foreign Direct Investment, Januar 2004
- Nr. 21: Bester, Helmut; Konrad, Kai A.: Easy Targets and the Timing of Conflict, Dezember 2003
- Nr. 20: Eckel, Carsten: Does globalization lead to specialization, November 2003
- Nr. 19: Ohr, Renate; Schmidt, André: Der Stabilitäts- und Wachstumspakt im Zielkonflikt zwischen fiskalischer Flexibilität und Glaubwürdigkeit: Ein Reform-ansatz unter Berücksichtigung konstitutionen- und institutionenökonomischer Aspekte, August 2003
- Nr. 18: Ruehmann, Peter: Der deutsche Arbeitsmarkt: Fehlentwicklungen, Ursachen und Reformansätze, August 2003
- Nr. 17: Suedekum, Jens: Subsidizing Education in the Economic Periphery: Another Pitfall of Regional Policies?, Januar 2003
- Nr. 16: Graf Lambsdorff, Johann; Schinke, Michael: Non-Benevolent Central Banks, Dezember 2002
- Nr. 15: Ziltener, Patrick: Wirtschaftliche Effekte des EU-Binnenmarktprogramms, November 2002
- Nr. 14: Haufler, Andreas; Wooton, Ian: Regional Tax Coordination and Foreign Direct Investment, November 2001
- Nr. 13: Schmidt, André: Non-Competition Factors in the European Competition Policy: The Necessity of Institutional Reforms, August 2001
- Nr. 12: Lewis, Mervyn K.: Risk Management in Public Private Partnerships, Juni 2001
- Nr. 11: Haaland, Jan I.; Wooton, Ian: Multinational Firms: Easy Come, Easy Go?, Mai 2001
- Nr. 10: Wilkens, Ingrid: Flexibilisierung der Arbeit in den Niederlanden: Die Entwicklung atypischer Beschäftigung unter Berücksichtigung der Frauenerwerbstätigkeit, Januar 2001
- Nr. 9: Graf Lambsdorff, Johann: How Corruption in Government Affects Public Welfare A Review of Theories, Januar 2001
- Nr. 8: Angermüller, Niels-Olaf: Währungskrisenmodelle aus neuerer Sicht, Oktober 2000
- Nr. 7: Nowak-Lehmann, Felicitas: Was there Endogenous Growth in Chile (1960-1998)? A Test of the AK model, Oktober 2000
- Nr. 6: Lunn, John; Steen, Todd P.: The Heterogeneity of Self-Employment: The Example of Asians in the United States, Juli 2000
- Nr. 5: Güßefeldt, Jörg; Streit, Clemens: Disparitäten regionalwirtschaftlicher Entwicklung in der EU, Mai 2000

- Nr. 4: Haufler, Andreas: Corporate Taxation, Profit Shifting, and the Efficiency of Public Input Provision, 1999
- Nr. 3: Rühmann, Peter: European Monetary Union and National Labour Markets, September 1999
- Nr. 2: Jarchow, Hans-Joachim: Eine offene Volkswirtschaft unter Berücksichtigung des Aktienmarktes, 1999
- Nr. 1: Padoa-Schioppa, Tommaso: Reflections on the Globalization and the Europeanization of the Economy, Juni 1999

Alle bisher erschienenen Diskussionspapiere zum Download finden Sie im Internet unter: <u>http://www.uni-goettingen.de/de/60920.html</u>.